Illusive Transparency? Evidence on Election Video Monitoring in Russia^{*}

Anastasiia Faikina[†]

June 10, 2022

CLICK HERE for the latest version

Abstract

Authoritarian regimes, like democracies, hold elections and often equip them with transparency-enhancing technologies. Why would autocrats want to hold or appear to hold transparent elections? This paper examines the impacts of broadcast election video monitoring in Russia. I exploit a discontinuity in the assignment of webcams to polling stations during the 2018 presidential election to estimate the causal effects on voting. Video monitoring reduces reported voter turnout by 5.2%and votes for the incumbent (autocrat) by 8.3%, suggesting a decrease in fraud. However, that decrease is partially offset by increased votes for the incumbent in neighboring unmonitored polling stations, indicating a displacement of fraud. To explore why autocrats implement video monitoring, I conducted a nationwide survey experiment before the 2019 local elections. Treated respondents were informed of video monitoring, which significantly increased the trust in elections and willingness to vote among those not previously aware of transparency technologies. Overall, these results suggest that video monitoring allows autocrats to improve citizens' attitudes by creating an illusion of transparency at a low cost in terms of net lost votes.

Keywords: Autocrats, Elections, Fraud, ICT, Video Monitoring

JEL Codes: D72, D73, P26

^{*}I am thankful to Eli Berman, Prashant Bharadwaj, Michael Callen, Gordon Dahl, Timothy Frye, Bruno Lopez-Videla, Andrea Matranga, Craig McIntosh, Karthik Muralidharan, Paul Niehaus, and participants of the North East Universities Development Consortium Conference at Boston University, Pacific Conference for Development Economics at UC Berkeley, Midwest International Economic Development Conference at Purdue University, CSAE Conference at Oxford University, Harriman/Jordan Workshop at Columbia University, APSA Annual Meeting, CREECA Young Scholars Conference at the University of Wisconsin-Madison, and Applied Workshops at UCSD for insightful and much-appreciated comments and suggestions. I gratefully acknowledge the financial support provided by the Harriman Institute at Columbia University, Carnegie Corporation of New York, UC Institute on Global Conflict and Cooperation, and Department of Economics at UCSD. The survey experiment is covered by UCSD IRB Protocol 191211 and pre-registered with the AEA as trial AEARCTR-0004635.

[†]Department of Economics, University of California San Diego.

I. Introduction

The last decade saw the backsliding of democracy worldwide, with a decline in the number of democracies and steady decreases in democratic scores across countries (Lührmann and Lindberg, 2019).¹ Nowadays, 43% of the world's population and 87% of the poor live in hybrid and authoritarian regimes.² These modern non-democratic regimes look different from those of the previous century. They maintain control less through the overt use of violence, repression, and indoctrination. Instead, they imitate core democratic institutions and hold elections at different levels of government (Guriev and Treisman, 2019*a*). More than 80% of non-democracies elect legislatures on a regular basis (Cruz, Keefer and Scartascini, 2021). Most of them invest in electoral transparency by inviting international observers, who monitor up to 84% of national non-democratic elections (Hyde and Marinov, 2012), and equipping polling places with transparencyenhancing technologies. Some examples include video monitoring, electronic vote-count and voting machines, and transparent ballot boxes. This raises the question: why would autocrats want to hold or appear to hold transparent elections?

This paper explores the effects of election video monitoring in Russia, a well-known authoritarian regime. This technology involves the placement of webcams inside polling stations, which then stream voting online (as depicted in Appendix Figure A.1). Studying election video monitoring is interesting for three reasons. First, economists model elections as an accountability mechanism that helps align the incentives of citizens and those of their government (Besley, 2006). However, these models cannot explain the need for transparent elections in autocracies, which do not always maximize social welfare. Second, elections represent critical junctures that can trigger the mobilization of the masses and democratization (Acemoglu and Robinson, 2012). Electoral irregularities facilitated revolutions in several post-communist countries, such as Georgia, Kyrgyzstan, Ukraine, and Serbia (Tucker, 2007; Bunce and Wolchik, 2011). Third, various states have increasingly adopted video monitoring, including Albania, Armenia, Azerbaijan, Georgia, India, Israel, and Ukraine.³ Many counties in the United States also install webcams in vote-counting centers.

The effects of video monitoring are ambiguous, especially in a non-democracy. On the one hand, it reduces the incentives of local authorities to commit election fraud inside

¹For example, Freedom House reports that the number of countries with aggregate *Freedom in the World* score declines outnumbered those with score gains every year for the last 15 years, and this gap has been increasing over time. Source: "Freedom in the World 2021: Democracy under Siege," freedomhouse.org

 $^{^{2}}$ Hybrid and authoritarian (non-democratic) regimes are states with Polity IV score of less than 6; *poor* are residents of low-income countries defined by the World Bank classification.

³In some countries, video monitoring raised concerns of ballot secrecy violation and voter intimidation. For example, Arab citizens claimed that the Israeli government used cameras against them as a voter-intimidation tactic in the 2019 legislative election. Source: "Israel Voting Cameras Lowered Arab Turnout, Netanyahu Backers Claim," published on April 10, 2019, by the *New York Times*: www.nytimes.com.

polling stations since citizens can detect irregularities online. In turn, the reduced scope for fraud can make local officials more accountable to citizens for spending on public goods. On the other hand, autocrats can use video monitoring to the regime's advantage. They can displace fraud to unmonitored polling stations or substitute it with forms of manipulation that are hard to observe on video. In such circumstances, video monitoring can mislead citizens and form biased beliefs about the fairness of elections, potentially leading to better attitudes toward the government. In the most extreme case, autocrats can use it to intimidate voters and coerce votes for the incumbent. This paper attempts to resolve the ambiguity in the effects of video monitoring using quasi-experimental evidence from the administrative data and experimental evidence from an original survey experiment.

I estimate the causal effects of video monitoring on election outcomes by exploiting a discontinuity in the assignment of webcams to polling stations in the 2018 presidential election in Russia. The Central Election Commission suggested installing webcams only inside polling stations with more than 1,000 registered voters. I employ a fuzzy regression discontinuity design around this cutoff and combine it with election data from 23,000 polling stations covering 25 million registered voters (23.3% of all voters). Specifically, I compare election outcomes at polling stations whose voter coverage is immediately above the cutoff for video monitoring to those at polling places just below the cutoff level.⁴

I present four main results from the regression discontinuity analysis. First, video monitoring reduces the reported voter turnout by 5.2%. It also leads to 8.3% fewer votes in favor of the incumbent without a corresponding change in the votes cast for other candidates.⁵ I show that these effects are consistent with a reduction in election fraud rather than a decrease in voters' willingness to vote. Using a standard elections forensic tool, I find that a large share of unmonitored polling stations experience abnormally high turnout and an abnormal increase in the incumbent's vote share; this pattern is commonly observed in countries with fraudulent elections. In contrast, video monitoring significantly reduces this share such that turnout and incumbent vote share resemble those for fair elections in democracies.

Second, I show that the direct effects of video monitoring are more prominent in rural areas known for higher levels of election fraud. Rural polling stations experience an 8.2% drop in votes cast for the incumbent. The estimate for urban areas is less precise, amounting to only a 5.7% reduction in votes. Nevertheless, only 14% of rural polling stations have webcams installed. This might indicate a decision not to install

⁴Not all eligible polling stations implemented video monitoring due to financial and legal constraints. Treatment-on-the-treated estimates adjust the differences for a 60-percentage-point gap in the implementation of video monitoring between the two groups.

⁵The results are robust to a series of robustness checks, including using different functional forms and bandwidths, and placebo tests using different elections and alternative thresholds.

video monitoring in rural areas so as to mitigate its effects on fraud overall. A back-ofthe-envelope calculation suggests that the universal adoption of video monitoring would result in a 55% higher loss in votes compared to the results under the actual system of implementation.

Third, I find that video monitoring displaces votes to neighboring unmonitored polling stations. Using a similar empirical strategy to that already described, I show that unmonitored polling stations located in close geographical proximity to monitored stations experience a spike in the share of votes cast for the incumbent. A rough estimate suggests that this spike offsets more than half of the direct effect of video monitoring on the incumbent's votes.

Fourth, I explore whether the reduced scope for election fraud disciplines local officials to provide more public goods. I instrument the intensity of video monitoring in each district with the share of polling stations above the cutoff in the bandwidth. I do not find evidence of differential changes in public spending or its re-allocation across sectors in districts with a higher intensity of video monitoring before the election. Hence, there are no observable accountability effects of monitoring on public goods spending.

I then explore *why* a non-democratic regime might introduce election video monitoring in the first place. I use an original survey experiment conducted before the 2019 local elections for a nationally representative sample of 1,097 prospective voters to adjudicate among the following theories.⁶ First, autocrats might concede some power to voters and invest in democratic institutions to prevent discontent (Acemoglu and Robinson, 2005). In particular, they might introduce video monitoring of elections to build the trust of citizens in electoral institutions and improve attitudes towards the regime (Fearon, 2011; Little, 2012). Second, autocrats might invest in fraud-reducing technologies to convince citizens of the regime's popularity and deter opponents (Egorov and Sonin, 2021). Third, autocrats might invest in video monitoring to intimidate voters and coerce support for the regime (Frye, Reuter and Szakonyi, 2014, 2019*a*).

The history of election video monitoring's introduction in Russia aligns with the first theory. In December 2011, Russia saw the largest protests since the fall of the USSR, triggered by numerous allegations of election fraud in the parliamentary elections. To convince voters of electoral integrity, the prime minister ordered the installation of cameras for future elections. The survey experiment further supports this theory.

The experiment provided a reminder about video monitoring of the upcoming local elections to a random half of respondents. I then inquired about respondents' voting intentions, perceptions about elections, and general views of democracy. I find that 60% of respondents had already been aware of webcams in the upcoming elections, 65% thought it was an effective policy, and 15% expressed their willingness to monitor elections online.

⁶Prospective voters are individuals who answered that their locality was going to have elections on the 2019 Single Voting Day.

Additionally, respondents who received a reminder about video monitoring were 16% more likely to trust the election results fully. This effect is mainly concentrated among respondents previously unaware of electoral transparency policies; this was proxied by the respondent's awareness of transparent ballot boxes. Those respondents who were unaware of transparency policies were 30% more likely to believe in the fairness of the results. They were 23% more likely to express an intention to turn out to vote and 33% more likely to vote for the incumbent (autocrat).⁷ Overall, information about video monitoring improved citizens' attitudes toward the regime.

The survey data does not provide strong support for the second theory, that video monitoring convinces citizens of the regime's popularity. The reminder about video monitoring has only a small and statistically insignificant effect on beliefs about others' support of the ruling party. While respondents in the control group, on average, believed that 47.8% of other voters support candidates from the ruling party, this number only increased to 49.8% in the treatment group.

The data also contradicts the third theory of voter intimidation. First, the findings of lower turnout and fewer votes cast for the incumbent are inconsistent with voter intimidation. One would expect to see the opposite pattern given the coercion of voters in non-democratic regimes. There are also no effects observed on the votes cast for other candidates, indicating no intimidation of opposition supporters. Second, according to the list experiment I conducted as part of the survey to analyze voter intimidation, 26% of respondents had fears that their employer or local authorities would know whether they voted or not.⁸ However, the level of voter intimidation does not significantly vary with a reminder about video monitoring.

To summarize, this paper makes several contributions to the existing literature. First, my findings contribute to the growing literature on the political economy of nondemocracies (see Egorov and Sonin (2020) for a review of that literature). In particular, my findings add to the political science literature on elections held by autocrats, reviewed in Gandhi and Lust-Okar (2009) and Gehlbach, Sonin and Svolik (2016). My results improve our understanding of why autocrats might want to appear to hold transparent elections. While the direct effects of video monitoring are consistent with improved electoral accountability, these results are deceptive given its implementation only at large polling stations and the subsequent displacement of votes. As a result, voters form overly optimistic beliefs about the effectiveness of video monitoring and increase their trust in electoral institutions. These conclusions are similar to those of Herron (2010) and

 $^{^7\}mathrm{The}$ latter is not statistically significant due to the smaller size of the sample of respondents willing to vote.

⁸The list experiment estimates sensitive beliefs without directly asking about them. All the respondents receive a list of statements related to the elections, and for a random half, an additional item about voter intimidation is included. They declare only the total number of statements with which they agree. The difference between the two treatment groups reveals the level of voter intimidation.

Sjoberg (2014), who study video monitoring of around 500 polling stations in Azerbaijan. I extend and deepen those insights by analyzing, for the first time, a nationwide video monitoring policy. The additional data, and the survey experiment, allow me to study vote displacement effects and distinguish between different theories.

My results contribute to other theories of why autocrats might want to hold transparent elections. They might use elections to collect information on the ruling party's support and performance of local officials (Cox, 2009; Malesky and Schuler, 2011; Miller, 2015; Martinez-Bravo et al., 2017). Autocrats may also want to appease the international community to gain legitimacy and receive foreign aid (Wright, 2009; Hyde and Marinov, 2014). However, as I discuss below, video monitoring is unlikely to be used for these reasons in Russia. Finally, my findings indicate potential trade-offs between reducing election fraud to improve the attitudes of the population at large and using fraud to deter opponents and project strength to allies (Simpser, 2013; Gehlbach and Simpser, 2015; Rozenas, 2016).

My conclusions are in line with the theory and cross-country empirical evidence of Guriev and Treisman (2019a, b), who show that modern autocrats manipulate information to signal the competence and benevolence of their regime. As my results indicate, autocrats use information about video monitoring to signal the fairness of elections, which improves attitudes toward the regime. This evidence contributes to the growing empirical literature on how autocrats manipulate citizens' beliefs. For example, Adena et al. (2015), King, Pan and Roberts (2017), and Chen and Yang (2019) show the effects of media tools, such as Nazi radio in Germany and Internet censorship in China, and Cantoni et al. (2017) show the effects of ideology promoted via a school curriculum.

Finally, my findings contribute to the literature on the impacts of transparency technologies on accountability and service delivery (Duflo, Hanna and Ryan, 2012; Lewis-Faupel et al., 2016; Muralidharan, Niehaus and Sukhtankar, 2016; Bossuroy, Delavallade and Pons, 2019; Banerjee et al., 2020; Muralidharan et al., 2021). Specifically, my findings are relevant to evaluations of technologies designed to improve electoral integrity (Callen and Long, 2015; Fujiwara, 2015). The direct effects of video monitoring suggest that this technology can improve electoral accountability when it is carefully implemented with this goal in mind. It thus offers a viable alternative to in-person election observation, which is proven to be effective (Hyde, 2007; Ichino and Schündeln, 2012; Casas, Diaz and Trindade, 2017; Enikolopov et al., 2013) but is often limited due to high costs. However, the indirect effects of video monitoring indicate that its use leaves room for manipulation and the displacement of votes from monitored to unmonitored polling stations. This result is consistent with the findings of Ichino and Schündeln (2012) and Asunka et al. (2017), who show similar displacement effects in the presence of traditional election observers in Ghana. It also accords with the result of Banerjee, Duflo and Glennerster (2008), who show that the local health administration adapted to the time-stamping machines monitoring nurse absenteeism in public health clinics in India. These results, taken together, indicate the need for more research on designing accountability-improving technologies that cannot be easily manipulated by authorities.

The remainder of the paper proceeds as follows. Section II sets out the institutional background, and Section III outlines the theoretical predictions. Section IV describes the data sources and introduces the primary empirical strategy. In Section V, I estimate the direct effects of video monitoring on election outcomes and discuss the mechanisms. Section VI evaluates the indirect impacts of video monitoring on the displacement of votes and public goods spending. In Section VII, I present evidence from my original survey experiment and use this to adjudicate among the theories of why autocrats invest in video monitoring. Section VIII concludes and makes policy recommendations.

II. Background

A. Elections in Russia

After growing democratic tendencies in the early 2000s, Russia has reverted to an autocratic state in the last decade.⁹ Nowadays, it classifies as an electoral authoritarian regime. These regimes imitate democratic institutions, including elections. Russia holds elections at all government levels, including federal, regional, and local elections. However, they are not entirely free and fair.

The electoral system of Russia consists of multiple levels. The first level is the Central Election Commission, an independent body responsible for conducting federal elections and overseeing regional and local elections. It consists of 15 members. The president, State Duma, and Federation Council of Russia each appoint five members. The second level consists of 85 regional election commissions, one per region. They help conduct federal election commissions formed in the territories of cities and districts (rayons) by the corresponding regional election commissions. The fourth, the lowest level, consists of polling stations responsible for the conduct of all elections. The territorial election commissions form them for a term of five years. Each polling station has 3–16 employees, depending on the number of registered voters.

The Criminal Code of Russia defines the following categories of electoral malfeasance: hindering exercise of voting rights or work of election commissions, falsification of election documents, illegal issue and receipt of ballot papers, and falsification of election results. The most severe punishment is imprisonment for a period of one to four years. In practice,

⁹The Polity IV score reached a value of +6 in the early 2000s. It reverted to +4 in 2007 and stayed at that level. This score belongs to the bottom 40% of countries in the democratic range and is similar to Guinea, Zimbabwe, Algeria, and Ethiopia.

prosecutions of electoral malfeasance are rare and usually penalized with monetary fines or community sentences. As a consequence, election fraud is widespread in Russia and well-documented by Myagkov, Ordeshook and Shakin (2009), Enikolopov et al. (2013), and Rundlett and Svolik (2016).

This paper studies the 2018 presidential and 2019 local elections. The presidential election has been contested by eight candidates, including the incumbent president (autocrat). The incumbent won with 76.7% of votes and voter turnout of 67.5%. The major international observation mission by the OSCE concluded that the election was held in an overly controlled environment, marked by continued pressure on critical voices and a lack of real competition. However, it also noted that the Central Election Commission administered the process efficiently and openly.¹⁰ The 2019 local elections selected representatives of different government levels, including parliamentary, gubernatorial (regional), and municipal. Candidates from the incumbent United Russia party won elections in most regions and municipalities.

B. Election Video Monitoring

In the last decade, the Central Election Commission of Russia invested substantial resources in technologies designed to improve the transparency and accountability of elections. It equipped polling stations with transparent ballot boxes, electronic vote counting and voting machines, video monitoring, and recently introduced electronic voting. The technology of video monitoring consists of placing webcams inside polling stations that stream the election day online. Russia is the largest implementer of video monitoring in the world, doing so on a national scale. Albania, Armenia, Azerbaijan, Georgia, India, Israel, and Ukraine have also implemented video monitoring in different years at a smaller scale. Many U.S. counties live-stream the vote-count process.

Russia first introduced video monitoring in the 2012 presidential election. The prime minister at the time ordered the installation of webcams in all polling stations in response to the large protests in December 2011, triggered by allegations of fraud in the 2011 parliamentary election.¹¹ As a result, 95% of all polling stations were equipped with webcams, around 90% of which streamed the election day online. Since then, regional and local election commissions have implemented video monitoring during lower-level elections at their discretion.

The 2018 presidential election brought back video monitoring on a national scale. The Central Election Commission implemented it at all territorial election commissions and

¹⁰The OSCE Office for Democratic Institutions and Human Rights: "Russian presidential election well administered, but characterized by restrictions on fundamental freedoms, lack of genuine competition, international observers say," www.osce.org

¹¹A VCIOM survey revealed that 45% of the respondents considered the 2011 parliamentary election as not free and fair, and 31% of the respondents did not trust the results. The Levada Center found that 46% of the respondents did not trust the results to some extent.

polling stations with more than 1,000 registered voters. Each monitored polling station had two webcams, a computer, and an Internet connection. The first webcam broadcasted the place of issuance of ballots to voters. The second webcam broadcasted the ballot boxes and voting booths.

Each camera streamed live video footage, which citizens could freely access on the website www.nashvybor2018.ru.^{12,13} The live stream covered the night before the election day (before 8:00 a.m.), voting (8:00 a.m.– 8:00 p.m.), and count of votes (after 8:00 p.m.). Voters could request access to a video recording from their polling station for three months after the election day. Presidential candidates could request access to any video recording during the same period. However, it was not easy to obtain such access in practice.

III. Theoretical Framework

Video monitoring can affect elections in two ways. First, it can change the behavior of local authorities, election administrators and politicians, by reducing the scope of election fraud. Second, it can impact voters who might change their voting behavior.

Similar to in-person observation, video monitoring reduces the incentives of local authorities to commit election fraud (see Hyde (2007) and Enikolopov et al. (2013) for the effectiveness of international and domestic monitors). Video monitoring can primarily prevent fraud easily observable on the video recording, such as ballot-box stuffing. It consists of casting ballots in place of registered voters who did not turn out to vote or bringing the same voters to multiple polling stations. Nevertheless, local officials can substitute observable fraud with other types, including voter registration and vote-count fraud, which are harder to identify on the video. They can also displace fraud to unmonitored polling stations. For example, Ichino and Schündeln (2012) find evidence of vote displacement in the presence of traditional election observers in Ghana.

If video monitoring successfully reduces election fraud, it might also trigger accountability effects. The classical principal-agent models consider elections as the primary accountability mechanism that helps align incentives of citizens and their government (Besley, 2006). Video monitoring might create additional incentives for politicians to improve the provision of public goods and services before the election. This would allow them to secure citizens' support of the incumbent and demonstrate loyalty to the regime.

Turning to voters, they receive information about video monitoring via media sources (TV, social media), signs at the polling stations, and a live-stream website. Since voters do not know about the monitoring status of their polling place before the election, they cannot change their registration for another polling station. However, they might change

 $^{^{12}}$ Appendix Figure A.1 presents screen shots of the website made on the election day.

¹³The Central Election Commission restricted access to the live video footage in the 2021 State Duma election. It is now available only at the special venues organized in the regional capitals.

their decision to vote or candidate choice on the election day if they receive information that their polling place is monitored. Moreover, they might change their voting behavior regardless of the monitoring status of their polling station if they react to the general information presented by the media.

On the one hand, video monitoring can trigger voter intimidation, which might have different effects on supporters and opponents of the regime in a non-democratic state. It can increase the turnout of government employees if they believe that their employer might see whether they voted or not. Frye, Reuter and Szakonyi (2014, 2019a,b) report widespread workplace mobilization in Russia and other non-democratic states. Video monitoring can also suppress the turnout of regime opponents if they believe it violates their vote secrecy. For example, Arab citizens claimed that the Israeli government used cameras against them as a voter-intimidation tactic in the 2019 legislative election.¹⁴

On the other hand, video monitoring can increase voters' trust in elections if they believe in its effectiveness in reducing election fraud. A representative poll of Russian citizens showed that 63% of respondents approved the introduction of video monitoring in 2012. The majority of respondents noted that it would reduce election fraud, promote fair and transparent elections, and allow to observe and document violations.¹⁵ Because of higher trust in election results, voters might be more likely to turn out to vote and even vote in favor of the regime. These effects should be larger among those not previously aware of other transparency initiatives.

Overall, the effects of video monitoring are ambiguous and depend on the relative magnitudes of these counterbalancing mechanisms. The following sections test outlined theoretical predictions using multiple data sources, empirical strategies, and an original survey experiment.

IV. Data and Empirical Strategy

This section starts with a description of each data source, including an original survey experiment conducted before the 2019 local elections. It proceeds with a description of the primary empirical strategy used to estimate the effects of video monitoring on voting. I describe the auxiliary empirical methods in the corresponding sections of results.

A. Data

Location of Video Monitoring. Estimating the effects of video monitoring requires knowledge on which polling stations had webcams. I collected this information from

 $^{^{14}}$ Source: "Israel Voting Cameras Lowered Arab Turnout, Netanyahu Backers Claim," published on April 10, 2019, by the *New York Times*: www.nytimes.com

¹⁵Source: Results of a representative survey on socio-economic and political topics conducted by Public Opinion Foundation (FOM) in January of 2012.

the official documents (decrees) issued by regional election commissions. This data was available on the websites of 26 out of 85 regions in Russia.¹⁶ These regions span more than 23,000 polling stations and cover more than 25 million (23.3%) registered voters. Appendix Figure A.2 shows that studied regions are spread geographically across the entire country. Appendix Table A.1 shows that studied regions have slightly lower average reported turnout and incumbent's vote share than the non-studied regions.

Election Outcomes. The primary administrative data source is the official results of the 2012 and 2018 presidential elections. I web-scraped this data from the website of the Central Election Commission of Russia in April–June 2018.¹⁷ This data contains the following information at the polling station level: the number of votes cast for each candidate, the number of registered voters, the number of voters who moved in and out of each polling station during the officially allowed period of 45 days before the election, and detailed information on ballots. The last includes the number of valid and invalid ballots and the number of ballots received by each polling station before the election day, cast inside and outside of each polling station, cast in stationary and carry boxes, and cast before the official voting day.

Polling Station Characteristics. I collected two sets of characteristics of the polling stations. First, I obtained the addresses of all the polling stations using the documents (decrees) and information on the websites of the regional election commissions. I geocoded the addresses of the polling stations using Google and Yandex maps. I then categorized the locations of all the polling stations into urban or rural areas using a text search algorithm.¹⁸ Second, I obtained the numbers and names of the polling station employees who worked in the 2018 presidential election for two regions in my sample.

Public Goods Spending. I collected data on local public procurement contracts to analyze the accountability effects of video monitoring on public goods spending. This data comes from the open registry of all the public procurement contracts in the country, available at www.zakupki.gov.ru. I obtained details on all the contracts signed four months before the 2018 presidential election (December 2017–March 2018) and the comparable period a year earlier (December 2016–March 2017). Each contract contains the following information: the code and name of each item, its value per unit, the number of units, the start and completion dates, the purchasing agency, and the funding source. I computed the total number and value of all the contracts per district (*rayon*), which include multiple territorial election commissions.

¹⁶The lack of this data for other regions can be due to poor website support, the negligence of bureaucrats, or to its intended unavailability.

¹⁷Different regions and outcomes were web-scraped on different dates.

¹⁸Russian addresses usually contain prefixes that distinguish settlement types. I define urban areas using addresses that include the following prefixes: "city," "urban settlement," and "urban-type settlement." Rural areas contain all other prefixes.

Survey Experiment. I also conducted an original survey experiment before the 2019 local elections to evaluate the effects of video monitoring on voter beliefs and behavior. It was carried out as part of the Omnibus survey conducted by the independent survey company Levada Market Research.^{19,20} The survey consisted of face-to-face interviews with 1,608 adults from 50 regions of Russia. The sample was stratified by gender, age, education, location, and municipality size using Russian statistical agency data. The resulting sample is representative of the Russian population with a margin of error of 3.4%. The survey included a block of questions about demographics, social status, and government approval, followed by a survey experiment to estimate the effects of video monitoring. The experiment was conducted for a subsample of 1,097 respondents who indicated that their locality was going to have elections on Single Voting Day (September 8, 2019). Section VII.A. describes the survey experiment design in detail.

B. Empirical Strategy

To estimate the causal effects of video monitoring, one ideally needs a random assignment of webcams to polling stations. In the absence of experimental variation, I exploit a quasi-experimental discontinuity in the implementation of video monitoring. In the 2018 presidential election, the Central Election Commission instructed regional election commissions to use a rule of 1,000 registered voters to define eligibility for video monitoring. Figure 1 shows a graphical representation of a discontinuity in the assignment of webcams resulting from the use of this rule. In total, 44% of the polling stations had video monitoring: 89% with more than 1,000 registered voters and 2% below this threshold.

I employ a fuzzy regression discontinuity design around the cutoff of 1,000 registered voters to estimate the causal effects. In particular, I compare the election outcomes at the polling stations right above the cutoff of 1,000 registered voters, which were eligible for video monitoring, to the results at the polling stations right below the cutoff, which did not qualify for it. The following equations summarize the empirical strategy:

$$m_s = \alpha_1 + \mathbb{1}\{v_s \ge c\} \left[f_r(v_s - c) + \delta \right] + \mathbb{1}\{v_s < c\} f_l(c - v_s) + \varepsilon_{1s}$$
(1)

$$y_s = \alpha_2 + \tau m_s + \mathbb{1}\{v_s \ge c\}g_r(v_s - c) + \mathbb{1}\{v_s < c\}g_l(c - v_s) + \varepsilon_{2s}$$
(2)

where m_s is a dummy variable equal to one if a polling station s was equipped with video monitoring, v_s is the number of registered voters, c is a cutoff of 1,000 registered voters, y_s is an election outcome, and f and g are unknown functions. The first-stage estimate δ estimates a jump in the probability of video monitoring at the threshold of 1,000

¹⁹Omnibus surveys allow administering nationally representative surveys at a reasonable cost in geographically spread-out countries like Russia by pooling questions from multiple researchers who share the costs.

 $^{^{20}\}mathrm{Levada}$ Market Research is known as the Levada Center in Russia.

registered voters. The second-stage estimate τ estimates the impact of video monitoring on the election outcome y.

The primary identifying assumption is no manipulation of the running variable, which determines eligibility for video monitoring. I choose as a running variable the number of registered voters 45 days before the election, i.e., before they could officially request to change their polling station. The regional election commissions decided where to install webcams before this date. Figure 2 plots the frequency of the running variable (Panel A) and McCrary density test (Panel B). There is no evidence of manipulation of voter registration at the time of webcam assignment (test statistic = -0.8, p-value = 0.42).

Another identifying assumption is no discontinuities in other polling station characteristics at the threshold of 1,000 registered voters that could have affected election outcomes. Appendix Figure A.3 shows that there is no discontinuity in the assignment of electronic vote-counting machines, another technology designed to improve electoral accountability. Appendix Figure A.4 shows for a subsample of two regions that there is no discontinuity in the number of polling station employees.²¹ Thus, there are no other discontinuities in the observable characteristics.

Another assumption required for a consistent estimation of the second stage is that video monitoring is the only channel causing differences in election outcomes. A potential concern is that the number of registered voters crossing the threshold can affect election outcomes for reasons unrelated to video monitoring. For example, voters at slightly bigger polling stations can have different preferences. To address this concern, I will report below (in Section V.B.) placebo tests using alternative thresholds and the preceding 2012 presidential election, which had cameras at all polling stations.

Two-stage least squares regression also requires a monotonicity assumption. The rule of 1,000 registered voters should not make any polling station less likely to install video monitoring. I do not directly observe what types of eligible polling stations did not have video monitoring. However, the regional election commissions made eligibility decisions based on the official guidelines issued by the Central Election Commission. The officials could not install cameras at the polling stations established in hospitals and other health organizations that have inpatient departments, detention centers, other places of temporary stay, military units, and polar stations. Most eligible polling stations that did not have video monitoring should be located in these restricted places.

V. Direct Effects

A. Implementation

The placement of video monitoring at the polling stations with more than 1,000 registered voters affects the scale of its implementation. Figure 1 and Table 1 show that 44%

²¹This data is difficult to obtain for all regions.

of all polling stations are equipped with webcams in 26 studied regions. They cover 73% of registered voters and are predominantly located in urban areas. Around 94% of eligible urban polling stations with more than 1,000 registered voters have video monitoring. The remaining 6% of eligible urban polling places are located in hospitals, detention centers, and other places restricted by law. At the same time, only 69% of eligible rural polling stations have webcams, reflecting either a lack of technical infrastructure or its intentional absence. Overall, video monitoring is present at 82% of urban and 14% of rural polling stations.

Panel A of Figure 3 shows a graphical representation of the first stage using the bandwidth of 400 registered voters and separate linear trends on each side of the cutoff. Polling stations with more than 1,000 registered voters have an approximately 60 percentage points higher probability of video monitoring than polling stations with fewer than 1,000 registered voters.

One can also notice that a few (84 or 1.4%) polling stations right below the cutoff experience a jump to an intermediate point of around 45 percentage points. There are strong reasons that this jump does not reflect manipulation. First, these polling stations receive video monitoring even though they are not eligible for it. Video monitoring is not a desirable outcome because it makes election fraud more difficult. Second, these polling stations constitute only 1.4% of all polling stations in the bandwidth and are spread across one-third of all territorial election commissions in 23 regions. If there is an intentional manipulation, one would expect these polling stations to be concentrated in a few geographical areas. Hence, this intermediate jump is likely a result of the non-strict implementation rather than intended manipulation. The regional election commissions may have also used a different period to determine eligibility for video monitoring compared to my analysis.²² Overall, the intermediate jump does not threaten the identification. However, it makes estimates noisier and biases them downward. To estimate a precise effect of video monitoring, hereafter, I remove polling stations with 975–1,000 registered voters that experience a jump from the primary analysis.

Panel A of Table 2 quantifies the first-stage estimates. Columns (1)-(3) show that polling stations with more than 1,000 registered voters have a 60 percentage points higher probability of video monitoring (significant at the 1% level, std. error = 2.8). The magnitude does not change with regional and territorial election commission fixed effects, which signals a similar implementation of video monitoring across different geographical units. Columns (4)–(5) indicate that implementation was less strict in rural areas, where slightly fewer eligible polling stations received video monitoring.

 $^{^{22}}$ The data on the number of registered voters is publicly available for two dates: the first day of the re-registration period (45 days before the election day) and the election day. I use the former as my running variable and a difference between the two dates as an outcome that quantifies re-registration.

B. Effects on Voting

Does video monitoring affect voting? We now turn to the first set of results. Panels B–D of Figure 3 show graphical representations of the reduced form estimates of the impact of video monitoring on election outcomes. Panel B shows an effect on reported turnout, defined as a ratio of the total cast votes over the number of registered voters. Polling stations with more than 1,000 registered voters eligible for video monitoring have an approximately two percentage points lower reported turnout. Hence, fewer votes were cast at the eligible polling stations. Panel C shows that the difference in reported turnout leads to approximately 22 fewer votes cast for the incumbent. At the same time, Panel D shows that there is no difference in the total votes cast for other candidates.

Panels B–D of Table 2 provide reduced form and second-stage estimates with and without region and election commission fixed effects. Column (3) of Panel B shows that polling stations equipped with video monitoring have 3.5 percentage points lower reported turnout (significant at the 1% level, std. error = 0.9). This effect corresponds to an approximately 5.2% decrease, given a mean turnout of 67.7%. Column (3) of Panels C and D shows that video monitoring leads to a reduction in the number of votes cast for the incumbent and no changes in the total votes cast for other candidates. Video-monitored polling stations have around 42 fewer votes cast for the incumbent compared to unmonitored polling stations (significant at 1% level, std. error = 9.4). It is equivalent to an approximately 8.3% decrease, given an average of 503 votes.

Appendix Table A.2 shows that these effects lead to an approximately one percentage point reduction in the incumbent's vote share and margin of victory.²³ Appendix Table A.3 shows no significant effects of video monitoring on the number of votes cast for each non-incumbent candidate.

These effects are not consistent with the hypothesis of voter intimidation. There is no effect on votes cast for the non-incumbent candidates who are more likely to be supported by opponents of the ruling party. Moreover, there is a negative effect on the votes cast for the incumbent. This result is inconsistent with the findings of Frye, Reuter and Szakonyi (2014) and Frye, Reuter and Szakonyi (2019*a*), who show that in non-democratic countries like Russia, local authorities and public employers induce citizens to vote in favor of the incumbent. Section VII. of this paper provides additional evidence from a survey experiment that information about video monitoring does not affect voter intimidation measured in a list experiment.

In contrast, these effects are consistent with a reduction in election fraud, which is shown to be more prevalent in rural areas of Russia (Myagkov and Ordeshook, 2008). Columns (4)–(5) of Table 2 test whether video monitoring has larger effects in rural areas. Column (4) shows that video monitoring leads to a significant 4.4 percentage points

²³Detecting impacts on these variables is challenging because video monitoring reduces the total votes (the denominator) and the votes cast for the incumbent (the numerator).

(std. error = 1.3) or 6.3% decrease in reported turnout at the rural polling stations. This effect corresponds to an 8.2% decrease in the number of votes cast for the incumbent. At the same time, video monitoring leads to an insignificant two percentage points (std. error = 1.3) or 3% decrease in reported turnout at the urban polling stations. This corresponds to a 5.7% decrease in the votes cast for the incumbent. Hence, the effects of video monitoring are larger in rural areas, more susceptible to election fraud, where it covers only 14% of rural polling stations (36% of registered voters). The latter might signal a strategic decision not to install video monitoring in rural areas to reduce its effective-ness in preventing fraud. A back-of-the-envelope calculation indicates that the universal adoption of video monitoring would result in a 55% higher loss in votes compared to the status quo implementation.²⁴

Appendix Section E. provides evidence from a series of placebo tests to check identification assumptions. First, Figure A.6 and Table A.8 show that there are no discontinuities in election outcomes at the threshold of 1,000 registered voters using data from the 2012 presidential election. That year, video monitoring covered almost all polling stations, irrespective of their sizes. Second, Table A.9 shows that there are no placebo effects using alternative thresholds of 850 and 1,150 registered voters.

Appendix Section F. shows robustness of the results to alternative specifications. Table A.10 shows that the estimates are only slightly smaller when not excluding intermediate points in the interval of 975–1,000 registered voters. Tables A.12 and A.11 show that the estimates are similar when using alternative functional forms (quadratic, cubic, and local linear) and bandwidths (300, 500, and optimal according to CCT), respectively.

C. Mechanisms

The previous section indicated that the effects of video monitoring on voting are consistent with a reduction in election fraud, which is more prominent in rural areas. This section tests further testable implications of that mechanism.

First, I use detailed records from the election day to understand whether the effects on votes come from regular, early, and mobile voting or voter registration. Table 3 shows that all polling stations receive approximately the same number of ballots before the election day (point estimate = -4.6, std. error = 9.3). However, monitored polling stations use 39 fewer ballots (std.error = 9.1), which is statistically equivalent to a reduction in votes cast for the incumbent. This difference comes from fewer ballots cast into a stationary ballot box located inside polling stations on election day. There are no significant differences in

 $^{^{24}}$ This calculation assumes an 8.2% decrease in votes at all rural polling stations and a 5.7% decrease in votes at all urban polling stations, irrespective of their size. This results in an additional 291 thousand votes reduction compared to the current loss of 533 thousand votes.

votes cast during early and mobile voting, collected in a carry box outside of the view of the webcam.²⁵ There are also no differences in the number of invalid and lost ballots.

Table 4 shows that there are, at most, small effects on voter registration and its possible manipulation.²⁶ The direction of the effects is consistent with monitored polling stations removing more existing voters and adding fewer new voters to the records. However, these effects are imprecise and small compared to the reduction in votes.

Second, I use election forensics tools that quantify election fraud by detecting abnormalities in the distribution of election results. I follow Klimek et al. (2012), Kobak, Shpilkin and Pshenichnikov (2016), and Lacasa and Fernández-Gracia (2019), who show that election fraud can be detected by a strong positive relationship between reported turnout and winner's vote share and a large mass of polling stations with abnormally high values of both variables.

Figure 4 shows a graphical representation of this tool for polling stations in the bandwidth of 400 registered voters. Polling places with fewer than 1,000 registered voters, ineligible for video monitoring, have a strong positive relationship between the reported turnout and incumbent's vote share. Moreover, there is a large mass of polling stations with abnormally high values (greater than 80%) of both variables. However, these abnormalities are much smaller at polling places with more than 1,000 registered voters, eligible for video monitoring.

Table 5 quantifies the effects of video monitoring on the probability of abnormally high (greater than 80%) values of turnout and incumbent's vote share. It shows that 17% of polling stations have high turnout, 27% have high incumbent's vote share, and 12.5% have high values of both variables. Video monitoring significantly reduces the probability of these abnormalities. For example, it decreases the likelihood of both abnormalities by 8.8 percentage points (significant at 1% level, std. error = 3.1). This effect is equivalent to a 70% reduction in the occurrence of abnormally high results. Moreover, columns (4) and (5) show that these abnormalities are four times more likely to happen at rural polling stations (18.7% versus 4.9% probability). Video monitoring eliminates them in urban areas and reduces them by half in rural areas.

These results are consistent with the hypothesis that video monitoring reduces election fraud, which is more prevalent in rural areas. In particular, they are compatible with a reduction in ballot-box stuffing (i.e., fraudulently casting ballots for the incumbent using those of registered voters who did not turn out to vote). In contrast, video monitoring does not affect another type of manipulation known as vote-count fraud (i.e., changing

 $^{^{25}}$ Voters can request early voting when they cannot visit an assigned polling station on election day because of vacations, business trips, work, study, public duties, or poor health conditions. Similarly, voters can request mobile voting at their residence because of poor health conditions, disabilities, or house arrest.

 $^{^{26}}$ Voters can officially change an automatically assigned polling station based on their residence to another polling station 45 days before the election.

the results during the vote count process). Election forensics literature measures this fraud by the probability of rounded results (i.e., results within a 0.05 margin of an integer number). Appendix Table A.5 shows that around 11% of polling stations experience this abnormality with no effects from video monitoring. Overall, these results indicate that video monitoring reduces ballot-box stuffing but does not eliminate all types of fraud.

Third, I show that the effects of video monitoring are stronger in election commissions with a higher winner's vote share in the 2012 and 2008 presidential elections. Based on the above election forensics literature, fraud is more likely in areas with higher turnout and winner's vote share. Appendix Table A.7 shows that video monitoring reduces the turnout by 8% (6.4%) and incumbent's votes by 11.2% (10.6%) in election commissions with the above-median winner's vote share in the 2018 (2012) presidential elections. In contrast, it decreases the turnout by only 1.6% (3.8%) and incumbent's votes by 4.8% (5.6%) in election commissions with the below-median winner's vote share. Thus, video monitoring is more effective in areas with a historically higher possibility of election fraud.

Finally, I provide additional evidence using the distribution of reported turnout on election day. If all voters have a similar distribution of times at which they vote, we would expect to see a gradual widening of the gap in turnout between monitored and unmonitored polling stations. However, Appendix Figure A.5 and Table A.6 indicate a different pattern. A one percentage point gap in reported turnout emerges at 10 a.m., when only 10% of the voters cast their votes. Then it does not significantly change for several hours and grows again after 3:00 p.m., when most citizens have already cast their votes. This pattern is consistent with a reduction in ballot box stuffing rather than changes in voter behavior. Election fraud is more likely to happen at the beginning and end of the election day when polling stations are not crowded. Additionally, election administrators have better information about how many people did not turn out to vote at the end of the day, whose ballots can be used for stuffing.

Overall, multiple forms of evidence strongly indicate that video monitoring reduced election fraud, specifically ballot box stuffing. In contrast, the results weigh against the voter intimidation hypothesis. The following sections explore how local authorities and citizens react to the presence of video monitoring.

VI. Indirect Effects

This section explores whether the negative effects of video monitoring on voting trigger changes in the behavior of local authorities, election administrators and politicians. First, I examine whether they displace fraud to neighboring unmonitored polling stations. Second, I estimate whether the reduced scope for election fraud makes local officials more accountable to citizens in public goods spending.

A. Displacement Effects

Only 44% of polling stations were equipped with video monitoring in the 2018 presidential election. Since it is difficult to commit fraud inside monitored polling stations, local authorities might displace it to unmonitored polling places. For example, Ichino and Schündeln (2012) show evidence of the displacement of fraud to unmonitored polling stations in the presence of traditional election observers in Ghana. Rundlett and Svolik (2016) show that in Russia and other non-democracies, election fraud is conducted not centrally by incumbents but rather locally by election commissions. Thus, the displacement of votes is more likely to happen inside the existing geographical boundaries.

To identify the displacement effects, one ideally wants to compare election results at unmonitored polling stations whose neighbors were randomly assigned video monitoring to voting outcomes at unmonitored polling places whose neighbors were randomly assigned to not be monitored. In the absence of random assignment, I develop an auxiliary empirical strategy based on quasi-random variation in the implementation of video monitoring. First, I take all unmonitored polling stations and their neighbors inside a geographical radius of three, five, or seven kilometers (depicted inside the red circle in Appendix Figure A.7). Second, I exclude unmonitored polling stations that have neighbors with more than 1,400 registered voters, the upper bandwidth boundary (shown in Panel C of Appendix Figure A.7). These neighbors were eligible for video monitoring, and hence, there could have been a displacement of fraud from them. However, they are outside of my identification window.²⁷ Third, I exclude neighbors with fewer than 600 registered voters, the lower bandwidth boundary (shown in red inside the red circle in Appendix Figure A.7). They were not eligible for video monitoring, and they are also outside of my identification window. Finally, I restrict the sample to unmonitored polling stations with the neighbors on only one side of the cutoff of 1,000 registered voters (i.e., all neighbors should have the same eligibility status). While these assumptions are quite restrictive, they allow me to causally estimate displacement effects.

To summarize, I compare the outcomes of unmonitored polling stations, which have all their neighbors in the bandwidth eligible for video monitoring (Panel A of Appendix Figure A.7), to the outcomes of unmonitored polling places, which have all their neighbors in the bandwidth ineligible for video monitoring (Panel B of Appendix Figure A.7). When several neighbors are on the same side of the cutoff, I take the average number of voters and the probability of video monitoring. The following equations summarize the empirical strategy:

$$\bar{m}_{s}^{ngh} = \alpha_{1} + \mathbb{1}\{\bar{v}_{s}^{ngh} \ge c\} \left[f_{r}(\bar{v}_{s}^{ngh} - c) + \delta \right] + \mathbb{1}\{\bar{v}_{s}^{ngh} < c\} f_{l}(c - \bar{v}_{s}^{ngh}) + \varepsilon_{1s}$$
(3)

$$y_s^{unm} = \alpha_2 + \tau \bar{m}_s^{ngh} + \mathbb{1}\{\bar{v}_s^{ngh} \ge c\}g_r(\bar{v}_s^{ngh} - c) + \mathbb{1}\{\bar{v}_s^{ngh} < c\}g_l(c - \bar{v}_s^{ngh}) + \varepsilon_{2s}$$
(4)

²⁷The neighboring polling stations were as good as random to receive video monitoring only if the number of registered voters was close to the eligibility cutoff of 1,000.

where y_s^{unm} is election outcome at the unmonitored polling station s, \bar{m}_s^{ngh} is the mean probability of video monitoring of its neighbors, \bar{v}_s^{ngh} is the mean number of registered voters of the neighbors, c is a cutoff of 1,000 registered voters, and f and g are unknown functions. The first-stage estimate δ estimates a jump in the mean probability of video monitoring among neighboring polling stations at the threshold of 1,000 registered voters. The second-stage estimate τ estimates the displacement effect from monitored neighbors to the unmonitored polling station s.

Figure 5 plots first-stage and reduced-form estimates of displacement effects in the radius of five kilometers. Panel A shows that neighboring polling stations with more than 1,000 registered voters have an approximately 60 percentage points higher probability of video monitoring. This estimate is consistent with the first-stage estimates in the main results. Panel B shows that unmonitored polling stations, which have neighbors with more than 1,000 registered voters, have approximately the same turnout. However, Panel C shows that they have a 3.7 percentage points higher incumbent's vote share.²⁸ Symmetrically, Panel D shows that they have a 3.7 percentage points lower share of votes cast for other candidates.

Table 6 provides the reduced and second-stage estimates for a radius of three, five, and seven kilometers. Panel B confirms that there is a slightly positive but not statistically significant effect on turnout (point estimate = 1.1, std. error = 2.8). Panel C shows that unmonitored polling stations with monitored neighbors have an approximately 6.7 percentage points higher incumbent's vote share compared to unmonitored places with unmonitored neighbors (significant at the 1% level, std. error = 2.3). This corresponds to an approximately 8.6% higher vote share, given a mean of 77.9%. Panel D shows that this increase leads to a symmetric decrease in the percentage of votes cast for other candidates. The displacement effects reach the highest value in the radius of five kilometers and decrease for longer radii. This is consistent with the idea that displacement effects are bounded by the existing geographical limits.

Appendix Table A.13 shows the robustness of these results to pooling all neighboring polling stations instead of taking the means. Appendix Table A.14 shows the robustness to restricting the sample to unmonitored polling stations that have a single neighbor in the bandwidth of 600–1,400 voters.

Overall, these effects are consistent with the displacement of fraud from monitored to neighboring unmonitored polling stations. The likely mechanism is vote-count fraud, given no significant effects on turnout. A back-of-the-envelope calculation suggests that it corresponds to approximately 25 votes for an average unmonitored polling station. In total, fraud displacement compensates for around 330 thousand votes, more than half of

 $^{^{28}}$ It is not possible to estimate displacement effects in levels because there is no clear relationship between votes cast at the unmonitored polling station and the size of its neighbors (the running variable).

the direct impact of video monitoring shown in Table $2.^{29}$

B. Accountability Effects

Fair elections are the primary accountability mechanism that helps align the incentives of citizens and those of their government (Besley, 2006). Thus, the net benefits of video monitoring depend on whether it makes authorities more accountable to citizens. Since video monitoring reduces the scope for election fraud, it can motivate politicians to provide more public goods before the election and increase incumbent's support.

To test this hypothesis, I collected data on all public procurement contracts signed at the local level. I aggregated the number and value of all the procured projects at the district level and matched them to the election records. To estimate the effects of video monitoring on public goods spending, I exploit the variation in the intensity of video monitoring across districts (*rayons*), shown in Figure A.8. Specifically, I examine whether districts with a higher share of video-monitored polling stations spend more on public goods before the election. Since the information about video monitoring is available only around four months before the election, I use a difference in total per capita spending between a period of four months before the 2018 presidential election (December 2017 – March 2018) and a corresponding period a year earlier (December 2016 – March 2017), which allows to absorb the time-invariant differences in spending across districts. To address the potential endogeneity of the intensity of video monitoring, I instrument it with the share of polling stations with more than 1,000 registered voters in the bandwidth. The following equations summarize the empirical strategy:

Share Monitored_{dr} =
$$\alpha + \delta \cdot$$
 Share Above Cutoff $BW_{dr} + \eta \cdot$ Share $BW_{dr} + \theta_r + \epsilon_{dr}$ (5)

$$\Delta \text{Spending}_{dr} = \beta + \tau \cdot \text{Share Monitored}_{dr} + \gamma \cdot \text{Share BW}_{dr} + \mu_r + \varepsilon_{dr} \tag{6}$$

where Share Monitored_{dr} is a share of polling stations equipped with video monitoring in the district d of the region r. Share Above Cutoff BW_{dr} is a share of polling stations with more than 1,000 registered voters in the bandwidth, an instrument for Share Monitored_{dr}. Share BW_{dr} is a share of polling stations in the bandwidth that controls for a non-random distribution of polling station sizes. The second-stage estimate τ estimates the impact of a 1% increase in the share of monitored polling stations on changes in district public spending.

The exclusion restriction requires that the instrument affects changes in public spending before the election only through the effect of video monitoring. While the assignment of webcams is quasi-experimental, the number of polling stations with 600-1,400 registered voters in a particular district may not be a random event. To address this concern, I control for the share of polling stations in the bandwidth, Share BW_{dr}, in both stages.

²⁹This calculation assumes a uniform displacement effect on the incumbent's vote share of 6.7 percentage points across all unmonitored polling stations.

Table 7 presents the results. Panel A shows that the instrument strongly predicts the total share of monitored polling stations. A 10 percentage points increase in the share of polling stations with more than 1,000 registered voters in the bandwidth of 400 registered voters leads to a 4.7 percentage points increase in the share of monitored polling stations (significant at the 1% level, std. error = 0.3).

Panels B and C show that, on average, districts start fewer projects and spend around RUB 300 (approximately \$4) less per capita before the election. However, they finish roughly the same number of projects with the same value by March. There are no significant effects of the intensity of video monitoring on changes in the number of projects and their value. The point estimates are -0.1 (std. error = 0.1) for the number of projects and -5.7 (std. error = 6.5) for the total value of projects in RUB. These estimates are close to zero and not statistically significant (confidence intervals include zero). Both rural and urban districts experience similar patterns.

Additionally, Appendix Table A.15 shows the heterogeneous effects by major spending sectors. On average, districts do not substantially change spending across sectors. There are also no effects of the intensity of video monitoring on spending changes. Thus, video monitoring does not lead to substitution effects across sectors.

Overall, the data show that video monitoring has no significant impact on public goods spending. Most likely, this is because it does not have large effects on the election results and leaves room for the displacement of votes to unmonitored polling stations.

VII. Why Video Monitoring?

The regression discontinuity analysis indicates that video monitoring reduces election fraud committed in favor of the incumbent. However, local authorities partially offset these effects by displacing fraud to neighboring unmonitored polling places. This raises the question: why would autocrats want to implement video monitoring? This section reports the results of an original survey experiment used to adjudicate between competing theories.

A. Improving Attitudes Toward the Regime

"Opposition will always indicate that the election was unfair. Always! ... To minimize the possibility of indicating that these [2012 presidential] or future elections will be dishonest or maybe dishonest, to knock out those who want to delegitimize the power in the country ... I propose and ask the Central Election Commission to set up webcams in all polling stations in the country ... so that the country sees what happens at a particular [ballot] box. To completely remove all falsifications on this matter."

> Prime Minister of Russia Annual Call-In Show December 15, 2011

The Russian prime minister (autocrat) gave this speech just a few days after the largest protests since the fall of the USSR, which were sparked by numerous allegations of election fraud in the 2011 legislative elections.³⁰ His statement indicates that the government introduced video monitoring to stop the spread of beliefs about the unfairness of elections that de-legitimized the regime and triggered protests.

Consistent with the fears of the Russian government, Tucker (2007) and Bunce and Wolchik (2011) show that electoral irregularities initiated revolutions in several postcommunist countries, including Georgia, Kyrgyzstan, Ukraine, and Serbia. A more recent example comes from Bolivia, where the incumbent president had to step down after losing control over violent protests sparked by allegations of election fraud.³¹

Acemoglu and Robinson (2005) model autocrats who concede to voters and invest in democratic institutions to prevent a revolution and regime change. Fearon (2011) theoretically shows the importance of investments in electoral institutions to prevent discontent, such as holding elections on a regular schedule and monitoring their conduct. Little (2012) takes it further by showing that manipulations will always occur in the equilibrium under a broad set of assumptions. However, citizens will know about the existence of fraud and discount the result, rendering fraud ineffective. As a result, incumbents will invest in the monitoring of elections to reduce visible, inefficient fraud.

Survey Experiment Design. To test these theories, I conducted a survey experiment for a nationally representative sample of 1,097 respondents approximately two weeks before the 2019 local elections.³² I first asked all respondents whether they knew about transparent ballot boxes, a proxy for awareness of transparency initiatives. I then introduced the priming treatment comprised of a reminder to a random half of respondents that many polling stations in the upcoming elections would have video monitoring. It also provided information that the government-run website was going to stream the election day online. The control group did not receive any information about video monitoring. After introducing the treatment, I inquired about respondents' voting intentions and beliefs, trust in elections, and general views on democracy. To estimate the effects on voter intimidation, I cross-randomized a list experiment, which included a veiled response treatment group indirectly asked about it. The table below summarizes the survey design:

Video Monitoring Priming Treatment

Do you know that many polling stations in the upcoming election will have webcams that will stream voting online? (Yes/No)

³⁰For instance, see the article "Russian election: Biggest protests since fall of USSR," published by BBC News on December 11, 2018: https://www.bbc.com/news/.

³¹For example, see the article "Bolivia's president resigns amid election-fraud allegations," published on November 10, 2019, by AP News: https://apnews.com/.

³²The survey was conducted on August 22–28; Single Voting Day was held on September 8.

The website nashvybor2019.ru will stream the election day online from all polling stations in the country equipped with webcams. You can visit nashvybor2019.ru on the election day and observe the voting at any polling station with video monitoring. Will you observe the election on nashvybor2019.ru? (Yes/No/Hard to tell)

	Video Monitoring Priming	
List Experiment	Control	Treatment
Direct Response	271	273
Veiled Response	288	265
Total	559	538

Results. I pool the list experiment groups together for all outcomes except voter intimidation. Equation (7) summarizes the empirical strategy:

$$Y_{ir} = \alpha + \tau \operatorname{Treat}_{ir} + \varepsilon_{ir} \tag{7}$$

where Y_{ir} is the outcome for respondent *i* from region *r*, and $Treat_{ir}$ is an indicator equal to one if the respondent received the priming treatment about video monitoring. Table 8 shows the balance of covariates between the treatment and control groups. In line with expectations, only 1 out of 21 outcomes is significant at the 5% significance level. Appendix Table A.16 shows the balance of covariates across both the priming and listing treatments.

The last part of Table 8 shows that 60% of treated respondents acknowledged that they were already aware of video monitoring in the upcoming local elections. Hence, the priming treatment provides new information only for 40% of respondents and reinforces existing knowledge for the remaining 60%. Moreover, 15% of respondents expressed their willingness to observe the elections online using video monitoring.

Figure 6 shows the effects of the priming treatment on the primary outcomes: trust in election results, voting intentions, and beliefs about others' voting behavior. First, a reminder about video monitoring increases respondents' trust in election results from 29.6% in the control group to 34.4% in the treatment group. This effect corresponds to a 4.8 percentage points or 16% increase in trust (significant at the 10% level, std. error = 2.7). It primarily happens in the subsample of respondents unaware of another transparency tool, transparent ballot boxes. The latter has a moderate correlation of 0.45 with awareness about video monitoring and thus serves as a proxy for general knowledge about electoral transparency policies. Only 22.6% of respondents unaware of transparent ballot boxes believe in fair election results in the control group. The priming treatment increases this share by 6.7 percentage points, a 30% increase (significant at the 10% level,

std. error = 3.6). In contrast, respondents who are already aware of transparent ballot boxes have a higher baseline level of trust in election results of 35.1%. The priming treatment slightly increases this share by 3.1 percentage points, an insignificant 9% increase (std. error = 3.3).

Second, a reminder about video monitoring also increases respondents' willingness to vote. In the entire sample, the share of respondents willing to vote increases from 60.8% to 63.6%. The effect is again mainly concentrated among unaware respondents. Around 49.2% of unaware respondents intend to vote in the control group. This share increases by 11.1 percentage points or 23% in the treatment group (significant at the 5% level, std. error = 4.5). In contrast, aware respondents have a higher baseline willingness to vote of 69.8%, which does not significantly change with the priming treatment (point estimate = -3.7, std. error = 4.2).

Finally, greater awareness of video monitoring leads to a higher willingness to vote for the incumbent United Russia party. About 25.8% of unaware respondents intend to vote for United Russia in the control group. This share increases by 8.5 percentage points or 33% in the treatment group (std. error = 5.9). On the other hand, aware respondents already had a higher baseline willingness to vote for United Russia of 44.5%, which is not significantly affected by the treatment (point estimate = -2.3, std. error = 4.9).

Table 9 shows the results for all pre-specified outcomes and includes p-values from Fisher's exact test and p-values adjusted for the multiple-hypotheses testing. The treatment effect on trust in election results is significant at the 10% level in both the full sample and the unaware subsample.³³ However, it does not survive the multiple-hypotheses testing because there is no effect on the second pre-specified outcome, beliefs that elections will lead to improvements. The treatment effect on the willingness to vote is significant at the 5% level and survives the multiple-hypotheses testing for the subsample of unaware respondents. The treatment effect on the willingness to vote for United Russia is border-line significant at the 10% level given the smaller number of respondents who answered this question. The total effect on the voting intentions index is 0.19 standard deviations, and it is significant at the 1% level for the unaware subsample.

In addition, Table 9 shows that there are no effects on the overall index of democratic values. However, the priming treatment leads to stronger beliefs that democracy has not yet been established in Russia. While 83% of respondents in the control group believe that Russia is at least partially a democracy, five percentage points fewer respondents believe so in the treatment group. These negative effects are present among both unaware and aware respondents. A possible explanation is that respondents might perceive some features of video monitoring as non-democratic. An alternative interpretation is that video

 $^{^{33}}$ The survey experiment was not powered to identify the effects at the 5% significance level. The existing capacities of the survey firm, their sampling of regions and settlements, and varying election cycles across geographical areas restricted the sample size to 1,097 respondents.

monitoring might remind voters that elections would remain unfair and non-democratic in its absence.

Appendix Figure A.9 and Table A.17 show that the priming treatment effects have larger magnitudes and higher precision in the subsample that excludes respondents residing in Moscow whose beliefs are harder to change. First, they have higher awareness of video monitoring because it has been present at all polling stations in most elections since 2012. Second, a larger share of residents have well-established views opposing the government.

Appendix Figures A.10 and A.11 show heterogeneous treatment effects on the primary outcomes using individual characteristics. Female, young, less educated, unemployed, and higher-income respondents have larger priming treatment effects on trust in the election results. However, these heterogeneous effects do not translate into a higher willingness to vote except for less educated respondents. In contrast, respondents who approve of the government, do not watch TV daily, and do not use the Internet have higher priming treatment effects on both the trust and willingness to vote.

Overall, these findings support the theory of autocrats using video monitoring to increase trust in elections and improve attitudes toward the regime. These effects are particularly strong among those who were unaware of other electoral transparency policies.

B. Alternative Theories

Convincing Voters of Government Popularity. Alternative theories rely on the premise that autocrats can invest in the conduct of elections to convince voters and elites of their popularity (Egorov and Sonin, 2021). They can use video monitoring to assure citizens that others genuinely support the regime by producing a picture of clean elections. I test this theory by measuring respondents' beliefs about what percent of the other respondents support candidates from the ruling United Russia party.

The last panel of Figure 6 shows that an average respondent in the control group believes that 47.8% of other respondents will vote for the candidates from the incumbent United Russia party. This share is ten percentage points higher than the actual number of respondents willing to vote for United Russia. Hence, respondents hold overly optimistic beliefs about the government's popularity. The priming treatment increases these beliefs by two percentage points (std. error = 1.4); this difference is small and not statistically significant.

Similarly, unaware respondents believe that 43.4% of others support the regime. The priming treatment increases this share by 3.3 percentage points (std. error = 2.2) or 7.6%. Aware respondents believe that 51.2% of others will vote for United Russia candidates. The priming treatment does not significantly change their beliefs (point estimate = 0.9, std. error = 2.0).

Overall, information about video monitoring slightly increases beliefs about the regime's popularity. However, these effects are not statistically significant and smaller in magnitude than the corresponding effects on trust in elections and respondents' own willingness to vote.

Intimidating Voters. Autocrats can also implement video monitoring to intimidate voters and coerce voting in favor of the regime. Frye, Reuter and Szakonyi (2014, 2019a,b) document that voter intimidation is a common practice in many non-democratic countries, including Russia. Even though webcams do not violate ballot secrecy, they capture whether people turn out to vote. As a result, citizens might feel intimidated that their employer or local authorities will know whether they voted. They might also have incorrect beliefs that webcams can capture their choice of candidates.

Several policy reports and media articles support these concerns. The Venice Commission and the OSCE Office jointly advised the government of Georgia to remove video monitoring. They wrote the following: "The use of recording devices in the polling station, even if it does not infringe on the secrecy of the ballot, may appear to do so and can also intimidate some voters. As such, this provision may have a chilling effect on suffrage rights, potentially leading to intimidation, fear, and coercion."³⁴ Similarly, media outlets reported that cameras inside polling stations intimidated Arab voters and lowered their turnout in the 2019 Israeli elections.³⁵

As I discussed in Section V., the direct effects of video monitoring on election outcomes are inconsistent with voter intimidation. First, video monitoring has a negative impact on the incumbent's votes. It contradicts the hypothesis of voter intimidation, which usually takes form of coercing votes in favor of the incumbent in non-democratic regimes. Second, video monitoring does not affect the votes cast for the opposition candidates. Hence, there is no suppression of opposition supporters.

I provide additional evidence against voter intimidation by interacting the priming treatment with the cross-randomized list experiment, a survey method used to mitigate respondents' social desirability bias when eliciting information about sensitive topics. The list experiment randomly split respondents into two groups and presented them with a list of statements. The direct response group received three statements about the election. The veiled response group received the same items plus an additional sensitive statement on voter intimidation (shown below). Afterward, all respondents were asked to indicate only the total number of statements with which they agreed. A difference between average responses in these two groups indicates the level of voter intimidation.

³⁴Source: Council of Europe, Venice Commission. Opinion No. 571/2010: Joint Opinion on the Election Code of Georgia. Strasbourg/Warsaw, June 9, 2010. Page 17: https://www.venice.coe.int/.

³⁵For example, see the article "Minority Arabs in Israel object to cameras at polling centers," published by Reuters on April 9, 2019: https://www.reuters.com/.

Cross-randomization allows me to estimate the overall level of voter intimidation and how it changes with a reminder about video monitoring. Equation (8) shows the empirical strategy:

$$\# \text{ Statements}_{ir} = \beta_0 + \beta_1 \operatorname{Treat}_{ir} + \beta_2 \operatorname{Sens}_{ir} + \beta_3 \operatorname{Treat}_{ir} \times \operatorname{Sens}_{ir} + \epsilon_{ir}$$
(8)

where Treat_{ir} is equal to one if respondent *i* from region *r* received the priming treatment about video monitoring, Sens_{ir} is equal to one if the respondent received a sensitive statement in the list experiment, and β_3 captures the effect of priming on voter intimidation.

List Experiment

You will now see several statements. You do not need to state whether you agree with each one or not. Instead, please indicate the total number of items you agree with after reading all of them.

- 1. I saw a campaign ad on TV or heard one on the radio.
- 2. My polling station is within walking distance of my house.
- 3. The electoral campaigns annoy me.
- 4 (Sensitive). My employer or local authorities might know whether I voted or not.

The first row of Table 10 shows the level of voter intimidation among respondents who did not receive a reminder about video monitoring. Around 26% of them feel intimidated by the perception that the local authorities or their employer might know whether they voted or not. This estimate is in line with the findings of Frye, Reuter and Szakonyi (2014, 2019a,b), who show that up to 17% of employees experience coercion in Russia. Moreover, while 30% of unaware respondents experience voter intimidation, only 22% of aware respondents experience it. This is a first piece of evidence against the theory of voter intimidation because awareness about transparent ballot boxes is predictive of knowledge about video monitoring.

The second piece of evidence comes from the interaction effect displayed in the third row of Table 10. The interaction effect is positive, but it is not statistically significant in the full sample (point estimate = 0.06, std. error = 0.11). This means that a reminder about video monitoring does not have a significant effect on voter intimidation. Moreover, there is a precisely zero effect for the respondents who were unaware of other transparency initiatives (point estimate = -0.00, std. error = 0.14).

Overall, the list experiment designed to measure voter intimidation shows no evidence of it being affected by video monitoring. These results are consistent with the conclusions presented earlier in section V..B., which shows no effect on the votes cast for opponents. Collecting Local Information. Autocrats can also invest in electoral transparency to gather information on the regime's true support and the performance of local authorities, resolving the so-called Dictator's Dilemma (Wintrobe, 2000). Miller (2015) provides cross-country evidence that negative electoral shocks help autocrats to adjust their public spending. Malesky and Schuler (2011) and Martinez-Bravo et al. (2017) show that elections help autocrats to monitor local officials.

However, it is unlikely that autocrats introduce video monitoring to collect local information. First, this theory provides a reasonable explanation for investing in the transparency of local elections. However, it does not thoroughly explain the need for video-monitoring the presidential election. As Egorov and Sonin (2021) note, a nationally representative poll would be a much cheaper and less risky way to gather the information.

Second, video monitoring does not entirely eliminate election fraud, which persists at unmonitored polling stations. If the government was interested in collecting accurate information, it would install video monitoring at all polling stations or use a technology that local authorities could not manipulate with other types of fraud.

Third, video monitoring does not discipline local politicians to be more accountable to citizens before the election. Hence, the central government cannot use it to identify and remove disloyal or poor-performing local officials.

Appeasing the International Community. Alternatively, non-democracies might invest in electoral transparency to appease the international community. First, international donors can put pressure on non-democratic regimes by making foreign aid conditional on fair elections (Wright, 2009). Second, international observers can serve the role of referees by revealing information on the fairness of elections and supporting or refuting the opposition's claims about election fraud (Hyde and Marinov, 2014).

However, Russia is unlikely to invest in the video monitoring of elections to attract foreign aid. It has one of the lowest debt-to-GDP ratios globally, equal to 17.8% in December 2020. Russia is also a foreign aid donor rather than its recipient. The Organization for Economic Co-operation and Development (OECD) reports that Russia provided USD 1.2 billion in official development assistance in 2019.³⁶

Similarly, international election observers do not consider video monitoring to resolve all electoral irregularities in Russia. The primary election observation mission organized by the OSCE Office for Democratic Institutions and Human Rights (OSCE/ODIHR) acknowledges the efforts of the Russian Central Election Commission to increase the transparency by installing web cameras in the 2012 presidential elections. However, it also highlights that "there are inherent limitations as to what web cameras can and cannot capture, and therefore, from an outset, they cannot be regarded as an ultimate safeguard against any possible manipulations."

³⁶Source: Organization for Economic Co-operation and Development (OECD), Development Co-Operation Profiles: www.oecd-ilibrary.org.

VIII. Conclusion

The last decade saw the backsliding of democracy worldwide. Modern authoritarian regimes mimic core democratic institutions. They regularly conduct elections at many levels of government and invest significant resources in improving their transparency. This paper studies why autocrats want to hold or appear to hold transparent elections by evaluating the effects of broadcast video monitoring of polling stations in Russia.

I estimate the impacts of this technology on voting by exploiting a discontinuity in the assignment of webcams to polling stations in the 2018 presidential election. I find that video monitoring reduces reported voter turnout by 5.2% and votes for the incumbent (autocrat) by 8.3%. Additional analysis suggests that these effects are consistent with a reduction in election fraud rather than an increase in voter intimidation. However, the decrease in the incumbent's votes is partially offset by increased election fraud in neighboring unmonitored polling stations. As a result, video monitoring does not lead to significant accountability effects on public goods spending.

To understand why autocrats implement video monitoring, I conducted a nationwide survey experiment before the 2019 Russian local elections. Treated respondents were informed of video monitoring, which significantly increased the trust in elections and willingness to vote among those not previously aware of transparency technologies. At the same time, information about video monitoring did not affect voter intimidation or beliefs about the support of the ruling party by others. Taken together, these results suggest that video monitoring allows autocrats to improve citizens' attitudes by creating an illusion of transparency at a low cost in terms of net lost votes.

These findings provide policy implications for international donors and civil society organizations that actively invest in the transparency of elections, including USAID, the Carter Center, the National Democratic Institute, and the ODIHR. My results show that authoritarian governments might use accountability-enhancing technologies to create a facade of transparency and influence citizens' beliefs and behavior. International organizations should carefully evaluate whether their activities, including election observation, help strengthen non-democratic regimes rather than promote democracy.

Similarly, international donors should assess the possible indirect effects of technologies designed to improve the transparency of other public institutions before sponsoring them to hybrid and non-democratic regimes. A mere introduction of digital technologies might not reach the intended goals in states with weak institutions. Such technologies might leave room for the manipulation of their use without proper sanctions.

Overall, these conclusions support the theoretical literature (Besley and Persson, 2011; Acemoglu and Robinson, 2012) and emerging policy views (World Bank, 2016) that states and international donors should first invest in building cohesive democratic institutions to reap the dividends of digital technologies for the general population. $\label{eq:constraint} Acknowledgments Acknowledgments$

Figures and Tables



Figure 1: Video Monitoring Implementation

Notes: This figure shows the implementation of video monitoring in the 2018 presidential election. Panel A plots the share of video-monitored polling stations over their sizes defined by the number of registered voters. Each point is an average of observations in bins of 100 registered voters. Five outlying polling stations with more than 3,300 registered voters are excluded for presentation purposes. The red vertical line depicts the cutoff of 1,000 registered voters, which defined eligibility for video monitoring. Panel B plots the share of polling stations and registered voters covered by video monitoring.

Figure 2: Manipulation Tests







Notes: This figure shows manipulation tests of the running variable. Panel A plots a histogram of the number of registered voters in bins of 100 for the full sample and urban and rural subsamples. Panel B plots a non-parametric density function of the number of registered voters on both sides of the cutoff. The manipulation density test based on McCrary (2008) fails to reject the null hypothesis of no discontinuity around the cutoff of 1,000 registered voters (test statistic = -0.8, p-value = 0.42).



Figure 3: Effects of Video Monitoring on Voting

Notes: This figure plots first-stage and reduced-form estimates of the impact of video monitoring on voting. Each point plots means of observations in bins of 25 registered voters. The solid lines plot predicted values and 95% confidence intervals of a linear regression estimated separately on each side of the cutoff within the bandwidth of 400 registered voters. For presentation purposes, detrended graphs difference out trends associated with a positive relationship between the number of votes and registered voters.



Figure 4: Effects of Video Monitoring on Fraud Indicators

Notes: This figure shows the association between eligibility for video monitoring and the share of polling stations with abnormally high values ($\geq 80\%$) of reported turnout and incumbent vote share, indicators of fraud based on election forensics literature (Klimek et al., 2012). Each point corresponds to a joint density of the reported turnout and incumbent vote share in bins of one percentage point. The sample includes polling stations in the bandwidth of 400 registered voters on both sides of the cutoff. Red lines depict the turnout and vote share of 80%. The Pearson's correlation between two variables equals 0.55 for polling stations with fewer than 1,000 registered voters (ineligible for video monitoring) and 0.41 for polling places above this cutoff (eligible for video monitoring).





Notes: This figure shows the displacement of votes to unmonitored polling stations. Panel A plots the share of unmonitored neighboring polling stations as a function of their number of registered voters. Panels B-D plot election outcomes at unmonitored polling stations as a function of the number of registered voters at neighboring polling places. This estimation (i) excludes unmonitored stations, which have neighbors with more than 1,400 voters (upper bandwidth boundary); (ii) excludes neighbors with fewer than 600 registered voters (lower bandwidth boundary); (iii) excludes neighbors outside of the 5 km radius from unmonitored polling stations; (iv) restricts the sample to unmonitored polling places, which have all neighbors on one side of the cutoff; (v) takes the average number of voters if there are several neighbors. Each point plots means of observations in bins of 25 registered voters. The solid lines plot predicted values and 95% confidence intervals of a linear regression estimated separately on each side of the cutoff within the bandwidth of 400 registered voters.


Figure 6: Survey Experiment: Effects of Priming on Voting and Attitudes

Notes: This figure plots the effects of the priming treatment, which informed respondents about video monitoring during the survey experiment. The sample consists of 1,097 respondents who answered that there were going to be elections in their locality. Each subfigure plots the effects for the full sample (All), the subsample of respondents who were not aware of transparent ballot boxes (Unaware), which serves as a proxy for awareness of video monitoring (correlation of 0.45), and the subsample of respondents who were aware of it (Aware). Brackets denote 90% confidence intervals using standard errors clustered by region.

Statistic:	Mean	Std.Dev.	Median	Min	Max	# Obs.
	(1)	(2)	(3)	(4)	(5)	(6)
Urban	0.45	0.50	0.00	0.00	1.00	23301
Registered Voters	1088.26	794.71	955.00	16.00	4172.00	23301
Reported Turnout	0.69	0.13	0.66	0.27	1.00	23299
Incumbent's Vote Share	0.76	0.08	0.76	0.06	1.00	23299
Incumbent's Vote Margin	0.63	0.14	0.64	-0.38	1.00	23299
Share of Votes Cast Inside	0.90	0.10	0.93	0.00	1.40	23299
Share of Votes Cast Outside	0.10	0.10	0.07	0.00	1.00	23299
Share of Early Votes	0.00	0.03	0.00	0.00	0.98	23299
Share of Invalid Votes	0.01	0.01	0.01	0.00	0.90	23299
Video-Monitored	0.44	0.50	0.00	0.00	1.00	23301
Video-Monitored (Urban)	0.82	0.39	1.00	0.00	1.00	10397
Video-Monitored (Urban, $\geq 1,000$ Voters)	0.94	0.24	1.00	0.00	1.00	8865
Video-Monitored (Urban, $< 1,000$ Voters)	0.11	0.31	0.00	0.00	1.00	1532
Video-Monitored (Rural)	0.14	0.35	0.00	0.00	1.00	12904
Video-Monitored (Rural, $\geq 1,000$ Voters)	0.69	0.46	1.00	0.00	1.00	2530
Video-Monitored (Rural, $< 1,000$ Voters)	0.01	0.08	0.00	0.00	1.00	10374

Table 1: Summary Statistics: 2018 Presidential Election

Notes: This table reports the summary statistics for the 2018 presidential election. The sample includes a universe of 23,301 inland polling stations in 26 studied regions, excluding 402 polling stations located on the vessels in navigation on the voting day. Election results are not available for two monitored polling stations, which canceled voting because of election fraud.

Sample:	Full	Full	Full	Rural	Urban		
	(1)	(2)	(3)	(4)	(5)		
Panel A: Share of Vide	eo-Monitored	Polling Stati	ons				
First Stage	0.595^{***}	0.598^{***}	0.595^{***}	0.549^{***}	0.623^{***}		
	[0.028]	[0.028]	[0.028]	[0.036]	[0.049]		
Outcome Mean	0.436	0.436	0.436	0.251	0.668		
	(0.496)	(0.496)	(0.496)	(0.434)	(0.471)		
Panel B: Effect on Reported Turnout							
Reduced Form (ITT)	-0.019**	-0.021***	-0.021***	-0.024***	-0.012		
	[0.009]	[0.007]	[0.005]	[0.007]	[0.008]		
Second Stage (2SLS)	-0.032**	-0.035***	-0.035***	-0.044***	-0.020		
<u> </u>	[0.015]	[0.011]	[0.009]	[0.013]	[0.013]		
Outcome Mean	0.677	0.677	0.677	0.704	0.644		
	(0.124)	(0.124)	(0.124)	(0.127)	(0.110)		
Panel C: Effect on Inc	umbent's Voi	tes					
Reduced Form (ITT)	-21.711**	-23.611***	-24.717***	-21.753***	-18.393**		
	[10.182]	[7.191]	[5.666]	[7.598]	[9.151]		
Second Stage (2SLS)	-36.476**	-39.491***	-41.572***	-39.656***	-29.516**		
	[17.069]	[11.861]	[9.357]	[13.670]	[14.654]		
Outcome Mean	502.540	502.540	502.540	488.609	520.949		
	(165.244)	(165.244)	(165.244)	(166.916)	(161.368)		
Panel D: Effect on Oth	ers' Votes						
Reduced Form (ITT)	-2.453	-2.345	-1.465	-4.654	-0.889		
	[3.681]	[3.084]	[2.847]	[3.214]	[5.869]		
Second Stage (2SLS)	-4.121	-3.923	-2.465	-8.484	-1.427		
- 、 ,	[6.167]	[5.163]	[4.794]	[5.888]	[9.433]		
Outcome Mean	152.477	152.477	152.477	131.774	178.102		
	(66.269)	(66.269)	(66.269)	(59.167)	(65.887)		
Region FEs	No	Yes	No	No	No		
Commission FEs	No	No	Yes	Yes	Yes		
# Observations	5721	5721	5721	3142	2515		

Table 2: Effects of Video Monitoring on Voting

Notes: This table reports first-stage estimates and effects of video monitoring on voting in the 2018 presidential election. Specifications use data at the polling station level, exclude intermediate data points in the interval of [975, 1000) registered voters, exclude election commission clusters with a single observation, and include separate linear trends on each cutoff side with triangular weights. The bandwidth is 400 registered voters on both sides of the cutoff. Standard errors clustered by election commission are reported in brackets. Standard deviations are reported in parentheses.

Outcome:			Num	ber of Ball	ots		
	Received	Unused	Inside	Outside	Early	Invalid	Lost
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Second Stage (ToT)	-4.597	39.254^{***}	-39.334***	-4.687	0.167	0.091	0.003
	[9.337]	[9.047]	[12.231]	[5.445]	[1.767]	[0.946]	[0.003]
Outcome Mean	933.093	270.173	601.534	59.820	1.564	7.629	0.003
	(235.671)	(148.486)	(190.300)	(53.167)	(22.556)	(6.919)	(0.058)
Region FEs	No	No	No	No	No	No	No
Commission FEs	Yes	Yes	Yes	Yes	Yes	Yes	Yes
# Observations	5721	5721	5721	5721	5721	5721	5721

Table 3: Mechanisms: Effects on Different Categories of Ballots

Notes: This table reports the effects of video monitoring on the number of ballots of different categories in the 2018 presidential election. Specifications use data at the polling station level, exclude intermediate data points in the interval of [975, 1000) registered voters, exclude election commission clusters with a single observation, and include separate linear trends on each cutoff side with triangular weights. The bandwidth is 400 registered voters on both sides of the cutoff. Standard errors clustered by election commission are reported in brackets. Standard deviations are reported in parentheses.

Outcome:		Number of Re	egistered Vo	ters
	Included	Asinh Included	Excluded	Asinh Excluded
	(1)	(2)	(3)	(4)
Second Stage (ToT)	-6.596	-0.021	2.775	0.104**
	[8.745]	[0.076]	[2.140]	[0.052]
Outcome Mean	47.195	4.093	53.604	4.548
	(68.772)	(0.907)	(28.026)	(0.561)
Region FEs	No	No	No	No
Commission FEs	Yes	Yes	Yes	Yes
# Observations	5721	5721	5721	5721

Table 4: Mechanisms: Effects on Voter Registration

Notes: This table reports the effects of video monitoring on the number of voters who changed their registration status for another polling place before the 2018 presidential election. Included (excluded) voters measure the number of voters added to (excluded from) voter registration records due to changes in the registration status. *Asinh* applies the inverse hyperbolic sine transformation to reduce the skewness of variables, which have a large share of zero values. Specifications use data at the polling station level, exclude intermediate data points in the interval of [975, 1000) registered voters, exclude election commission clusters with a single observation, and include separate linear trends on each cutoff side with triangular weights. The bandwidth is 400 registered voters on both sides of the cutoff. Standard errors clustered by election commission are reported in brackets. Standard deviations are reported in parentheses.

Outcome:	Share of Po	lling Stations v	with Abnormal	ly High ($\geq 80\%$	%) Values of		
	Turnout Inc. Vote Turnout and Inc. Vote Share						
		Share					
Sample:	Full	Full	Full	Rural	Urban		
	(1)	(2)	(3)	(4)	(5)		
Second Stage (ToT)	-0.098***	-0.111**	-0.088***	-0.099**	-0.061*		
	[0.030]	[0.043]	[0.031]	[0.048]	[0.036]		
Outcome Mean	0.172	0.269	0.125	0.187	0.049		
	(0.377)	(0.443)	(0.331)	(0.390)	(0.215)		
Region FEs	No	No	No	No	No		
Commission FEs	Yes	Yes	Yes	Yes	Yes		
# Observations	5721	5721	5721	3142	2515		

Table 5: Mechanisms: Effects on Fraud Indicators

Notes: This table reports the effects of video monitoring on the share of polling stations with abnormally high values ($\geq 80\%$) of reported turnout and incumbent vote share, fraud indicators in election forensics literature (Klimek et al., 2012). Specifications use data at the polling station level, exclude intermediate data points in the interval of [975, 1000) registered voters, exclude election commission clusters with a single observation, and include separate linear trends on each cutoff side with triangular weights. The bandwidth is 400 registered voters on both sides of the cutoff. Standard errors clustered by election commission are reported in brackets. Standard deviations are reported in parentheses.

Radius:	3]	ĸm	5]	km	7	km
	(1)	(2)	(3)	(4)	(5)	(6)
Panel A: Share of Vide	eo-Monitored	Neighboring	Polling Statio	ons		
First Stage	0.501^{***}	0.527^{***}	0.577^{***}	0.582^{***}	0.613^{***}	0.617^{***}
	[0.077]	[0.075]	[0.071]	[0.070]	[0.059]	[0.059]
Outcome Mean	0.203	0.203	0.185	0.185	0.166	0.166
	(0.384)	(0.384)	(0.374)	(0.374)	(0.360)	(0.360)
Panel B: Effect on Rep	orted Turnov	ıt				
Reduced Form (ITT)	0.006	-0.001	0.004	0.006	0.013	0.008
	[0.026]	[0.019]	[0.023]	[0.016]	[0.021]	[0.015]
Second Stage (2SLS)	0.012	-0.003	0.007	0.011	0.021	0.013
	[0.052]	[0.036]	[0.039]	[0.028]	[0.033]	[0.025]
Outcome Mean	0.712	0.712	0.727	0.727	0.739	0.739
	(0.134)	(0.134)	(0.131)	(0.131)	(0.132)	(0.132)
Panel C: Effect on Inc	umbent's Vot	e Share				
Reduced Form (ITT)	0.029	0.023^{*}	0.037^{**}	0.039^{***}	0.038**	0.036^{***}
	[0.022]	[0.013]	[0.019]	[0.013]	[0.016]	[0.011]
Second Stage $(2SLS)$	0.058	0.043^{*}	0.065^{*}	0.067^{***}	0.063^{**}	0.058^{***}
	[0.045]	[0.025]	[0.034]	[0.023]	[0.026]	[0.019]
Outcome Mean	0.764	0.764	0.779	0.779	0.788	0.788
	(0.095)	(0.095)	(0.095)	(0.095)	(0.093)	(0.093)
Panel D: Effect on Oth	ners' Vote Sh	are				
Reduced Form (ITT)	-0.029	-0.022*	-0.037**	-0.038***	-0.038**	-0.035***
	[0.021]	[0.013]	[0.018]	[0.012]	[0.015]	[0.011]
Second Stage $(2SLS)$	-0.058	-0.043*	-0.064**	-0.065***	-0.062**	-0.057***
	[0.043]	[0.025]	[0.032]	[0.023]	[0.025]	[0.019]
Outcome Mean	0.225	0.225	0.210	0.210	0.202	0.202
	(0.092)	(0.092)	(0.092)	(0.092)	(0.090)	(0.090)
Region FEs	No	Yes	No	Yes	No	Yes
Commission FEs	No	No	No	No	No	No
# Observations	1095	1095	1538	1538	2061	2061

Table 6: Effects on Vote Displacement

Notes: This table reports vote displacement effects to unmonitored polling stations in the 2018 presidential election. This estimation (i) excludes unmonitored stations, which have neighbors with more than 1,400 voters (upper bandwidth boundary); (ii) excludes neighbors with fewer than 600 registered voters (lower bandwidth boundary); (iii) excludes neighbors outside of the specified radius from unmonitored polling stations; (iv) restricts the sample to unmonitored polling places, which have neighbors on one side of the cutoff; (v) takes the average number of voters if there are several neighbors. Specifications use data at the polling station level, exclude intermediate data points in the interval of [975, 1000) registered voters, and include separate linear trends on each cutoff side with triangular weights. The bandwidth is 400 registered voters on both sides of the cutoff. Standard errors clustered by election commission are reported in brackets. Standard deviations are reported in parentheses.

Projects:	S	tarted by Ma	rch	Started	and Finished	by March
Sample:	Full	Rural	Urban	Full	Rural	Urban
	(1)	(2)	(3)	(4)	(5)	(6)
Panel A: Share of Vi	deo- $Monitore$	d Polling Stat	tions (%)			
First Stage	0.471^{***}	0.143^{***}	0.555^{***}	0.471^{***}	0.143^{***}	0.555^{***}
	[0.033]	[0.021]	[0.047]	[0.033]	[0.021]	[0.047]
Outcome Mean	32.493	16.526	51.614	32.493	16.526	51.614
	(28.283)	(13.720)	(29.132)	(28.283)	(13.720)	(29.132)
Panel B: Change in t	he Number of	f Projects				
Second Stage (ToT)	-0.103	0.011	-0.218	-0.027	-0.142	0.001
	[0.124]	[0.469]	[0.196]	[0.034]	[0.143]	[0.050]
Outcome Mean	-10.691	-7.228	-15.378	-0.992	-0.170	-2.064
	(40.093)	(30.002)	(49.485)	(10.860)	(9.180)	(12.563)
Panel C: Change in the Total Value of Projects per Capita						
Second Stage (ToT)	-5.681	0.673	-11.180	-1.814	-8.499	-0.378
	[6.477]	[28.861]	[8.911]	[1.295]	[6.579]	[1.567]
Outcome Mean	-321.638	-309.724	-359.049	0.756	13.414	-24.203
	(2086.590)	(1930.417)	(2254.078)	(406.898)	(405.370)	(401.114)
Region FEs	Yes	Yes	Yes	Yes	Yes	Yes
First Stage F-Stat	207.629	45.687	139.529	207.629	45.687	139.529
# Observations	596	324	267	596	324	267

Table 7: Effects on Changes in Public Goods Spending

Notes: This table reports the effects of video monitoring on changes in district public goods spending before the 2018 presidential election. Changes in the number of projects and their total value are defined by the difference between December 2017 – March 2018 (four months before the election) and December 2016 – March 2017 (a corresponding period a year earlier), with winsorized values in the bottom- and top-2%. Specifications use data at the district level, instrument the share of video-monitored polling stations with the percentage of polling places above the cutoff in the bandwidth of 400 registered voters, and control for the share of polling stations in the bandwidth. Standard errors clustered by region are reported in brackets. Standard deviations are reported in parentheses.

Statistic:	Mean	Mean	Mean	Difference	P-value	# Obs.
-	Full	Control	Treat	Treat-Control		
	(1)	(2)	(3)	(4)	(5)	(6)
Geographical Area				0.04		100-
Urban (=1)	0.78	0.78	0.77	-0.01	0.68	1097
	(0.42)	[0.04]	[0.04]	[0.02]		
Moscow (=1)	0.13	0.13	0.12	-0.01	0.32	1097
	(0.33)	[0.12]	[0.11]	[0.01]		100-
St. Petersburg $(=1)$	0.05	0.04	0.06	0.02	0.29	1097
	(0.22)	[0.04]	[0.06]	[0.02]		
Rural $(=1)$	0.22	0.22	0.23	0.01	0.68	1097
	(0.42)	[0.04]	[0.04]	[0.02]		
Individual Characteristics	0 50			0.00	0.00	1005
Female $(=1)$	0.56	0.57	0.55	-0.02	0.63	1097
A	(0.50)	[0.02]	[0.02]	[0.03]	0.10	1007
Age	46.21	45.56	46.88	1.33	0.19	1097
	(16.50)	[0.68]	[0.73]	[1.00]	0.67	1007
Incomplete Secondary School $(=1)$	0.05	0.04	0.05	0.01	0.67	1097
	(0.21)	[0.01]	[0.01]	[0.01]	0.71	1007
Secondary or Vocational Education $(=1)$	0.64	0.65	0.64	-0.01	0.71	1097
	(0.48)	[0.03]	[0.02]	[0.03]	0.04	1007
Higner Education $(=1)$	(0.31)	0.31	0.31	0.00	0.84	1097
	(0.46)	[0.04]	[0.03]	[0.02]	0.00	1007
Employed $(=1)$	(0.60)	0.59	00.0	0.02	0.00	1097
\mathbf{D} (1)	(0.49)	[0.03]	[0.02]	[0.03]	0.79	1007
Retired $(=1)$	(0.26)	0.25	0.20	0.01	0.73	1097
T T I	(0.44)	[0.02]	[0.02]	[0.03]	0 50	1000
Income Level	3.18	3.17	3.20 [0.06]	0.04	0.58	1096
Madia Communitien	(1.02)	[0.07]	[0.06]	[0.00]		
D : L t t U (1)	0.00	0.69	0 50	0.04	0.10	1007
Daily Internet User $(=1)$	(0.60)	0.62	0.58	-0.04	0.16	1097
Internet New Here (1)	(0.49)	[0.03]	[0.03]	[0.03]	0.95	1007
Internet Non-User $(=1)$	(0.22)	[0.02]	0.22	00.0	0.85	1097
$D_{\rm c}$: $T_{\rm V}$ $U_{\rm corr}$ (1)	(0.41)	[0.02]	[0.03]	[0.02]	0.41	1007
Daily $1 \vee \text{User} (=1)$	(0.07)	0.00	0.00	0.05	0.41	1097
TV Non Ligon (-1)	(0.47)	[0.05]	[0.02]	[0.05]	0.00	1007
$1 \vee \text{Non-Oser} (=1)$	(0.08)	[0.09]	0.08	0.00	0.82	1097
Covennment Approval	(0.28)	[0.02]	[0.01]	[0.02]		
Approved President's Work (-1)	0.66	0.64	0.60	0.06	0.04	1080
Approves 1 resident s work (-1)	(0.47)	[0.02]	[0.09]	[0.02]	0.04	1080
Approves State Duma's Work (-1)	(0.47)	[0.03]	0.30	[0.03]	0.55	1066
Approves State Dunia's Work (-1)	(0.38)	[0.03]	[0.03]	[0.04]	0.55	1000
Approves Covernment's Work (-1)	(0.45)	[0.05]	0.46	0.04	0.75	1074
Approves Government's Work (-1)	(0.45)	[0.03]	[0.03]	[0.03]	0.75	1074
Approves Apy (-1)	0.71	0.70	$\begin{bmatrix} 0.05 \end{bmatrix}$	0.01	0.70	1094
Approves Any (=1)	(0.46)	[0.03]	[0.02]	[0.03]	0.10	1054
Transparency Technologies	(0.40)	[0.00]	[0.02]	[0.00]		
Aware of Transparent Ballot Boxes (-1)	0.56	0.56	0.56	0.00	0.89	1097
	(0.50)	[0.03]	[0.02]	[0.03]	0.00	1001
Aware of Video Monitoring $(=1)$	0.60	-	0.60	-	_	538
read of the monitoring (-1)	(0.49)		[0.02]			000
Will Observe Election Online $(=1)$	0.15	_	0.15	_	_	523
	(0.36)		[0.02]			010

 Table 8: Survey Experiment: Summary Statistics and Balance Test

Notes: This table reports summary statistics and balance test of random assignment of 1,097 respondents to priming treatment in the survey experiment. Column (1) reports the mean of each variable, with standard deviations in parentheses, for the full sample. Columns (2) and (3) report the mean of each variable, with standard errors clustered by region in brackets, for each experimental condition. Column (5) documents the difference in means between treatment and control groups, with standard errors clustered by region in brackets. Column (6) reports the p-value of a t-test of equality of means.

nd Attitudes	
Voting a	
ng on	
of Primi	
Effects	
periment:	
EX	
Survey	
Table 9:	

Outcomo.			U V					IInouro	04				Autorio		
Outcome.	Control	Treatment	ntr. D_trilert_D	Fisher's	MHT	Control	Treatment	D_walite	LC Fichar's	MHT	Control	Treatment	D_wate	Ficher's	MHT
	Mean	Effect		P-value	Q-value	Mean	Effect	ONTRA- T	P-value	Q-value	Mean	Effect		P-value	Q-value
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)	(13)	(14)	(15)
Voting/Supporting United Russia															
Will Vote $(=1)$	0.61	0.03	0.40	0.35	.93	0.49	0.11^{**}	0.02	0.02	.05	0.70	-0.04	0.37	0.34	1
	(0.49)	[0.03]				(0.50)	[0.04]				(0.46)	[0.04]			
Will Vote for United Russia $(=1)$	0.38	0.01	0.80	0.80	1	0.26	0.08	0.16	0.14	.12	0.45	-0.02	0.64	0.63	1
	(0.49)	[0.04]				(0.44)	[0.06]				(0.50)	[0.05]			
% Others Voting for United Russia	47.83	1.98	0.16	0.23	.93	43.36	3.33	0.14	0.18	.12	51.17	0.86	0.67	0.70	1
	(24.77)	[1.39]				(22.70)	[2.21]				(25.76)	[2.03]			
Trust in Election															
Trust in Election Results $(=1)$	0.30	0.05^{*}	0.09	0.10	.21	0.23	0.07^{*}	0.07	0.11	.18	0.35	0.03	0.36	0.45	.56
	(0.46)	[0.03]				(0.42)	[0.04]				(0.48)	[0.03]			
Election Will Lead to Improvements $(=1)$	0.37	-0.02	0.33	0.41	.21	0.31	-0.02	0.62	0.71	.45	0.42	-0.03	0.36	0.43	.56
	(0.48)	[0.03]				(0.46)	[0.03]				(0.49)	[0.04]			
Democratic Values															
Russia is a Democracy $(=1)$	0.83	-0.05**	0.03	0.03	.08	0.81	-0.05	0.15	0.18	.37	0.84	-0.05	0.10	0.11	.46
	(0.38)	[0.02]				(0.39)	[0.04]				(0.37)	[0.03]			
# Democratic Values Agreed	2.35	-0.04	0.65	0.70	.76	2.43	-0.18	0.18	0.23	.37	2.30	0.07	0.48	0.55	.93
	(1.57)	[0.08]				(1.66)	[0.13]				(1.50)	[0.10]			
Tax Compliance is Important $(=1)$	0.65	0.02	0.45	0.54	.76	0.64	0.04	0.30	0.41	.37	0.66	0.00	0.93	0.95	1
	(0.48)	[0.02]				(0.48)	[0.04]				(0.47)	[0.04]			
Indices															
Voting/Supporting United Russia Index	0.00 (0.59)	0.05 [0.03]	0.12	0.14	.32	0.00 (0.56)	0.19^{***} $[0.04]$	0.00	0.00	0	0.00 (0.59)	-0.03 [0.05]	0.55	0.51	.83
Truet in Flootion Index	0.00	[]	0 53	0 78	с С	000	900	86 U	0.30	11	000		0.08	00.0	00
	(0.79)	[0.04]	000	0000	00.	(0.77)	[0.06]	1.0	0	1	(0.80)	[0.06]		22.0	2
Democratic Values Index	0.00	-0.04	0.14	0.21	.32	0.00	-0.06	0.24	0.26	.41	0.00	-0.03	0.46	0.52	.83
	(0.56)	[0.03]				(0.53)	[0.05]				(0.59)	[0.04]			
* $p < 0.10$, ** $p < 0.05$, *** $p < 0.01$ Notes: This table reports the results of the survey.	exneriment	hased on the n	me-analysis	nlan. Colum	ns (1)_(5) renor	t the results	or the full can	of 1 005	' respondent	Columns (6)_	(11) pus (01)-	-(15) renort th	a results ser	arately for s	hsamles

Note: This that reports the results of the sturvey experiment based on the pre-marysrs plan. Columns (1)–(5) report the results for the rull sample of 1,097 respondents. Columns (0)–(10) and (11)–(15) report the results separately for stubstanples N(t) = 100 mm s(1), N(t) = 100 mm s(

Outcome:		Sample	
# Statements Agreed	Full	Unaware	Aware
	(1)	(2)	(3)
Sensitive Statement $(=1)$	0.26***	0.30***	0.22**
	[0.07]	[0.10]	[0.09]
Priming Treatment $(=1)$	0.02	0.07	-0.01
	[0.06]	[0.10]	[0.09]
Sensitive x Priming $(=1)$	0.06	-0.00	0.10
	[0.11]	[0.14]	[0.14]
No Sens., Control Mean	1.51	1.37	1.62
	(0.79)	(0.78)	(0.79)
# Observations	1062	464	598

Table 10: Survey Experiment: Effects of Priming on Voter Intimidation

Notes: This table reports the effects of the priming treatment on voter intimidation using a listing experiment. The first row estimates the level of voter intimidation in the priming control group (the impact of a sensitive item on the number of statements respondents agree with). The second row estimates the effect of the priming treatment on the number of statements in the listing control group. The third row estimates the priming treatment effect on voter intimidation (the interaction effect of two treatments). Column (1) shows estimates for the full sample. Columns (2)–(3) document estimates for subsamples split by awareness about transparent ballot boxes, a proxy for awareness about video monitoring (correlation = 0.45). Standard errors clustered by region are reported in brackets. Standard deviations are reported in parentheses.

References

- Acemoglu, Daron, and James A Robinson. 2005. Economic Origins of Dictatorship and Democracy. Cambridge University Press.
- Acemoglu, Daron, and James A Robinson. 2012. Why Nations Fail: The Origins of Power, Prosperity, and Poverty. Crown Books.
- Adena, Maja, Ruben Enikolopov, Maria Petrova, Veronica Santarosa, and Ekaterina Zhuravskaya. 2015. "Radio and the Rise of the Nazis in Prewar Germany." *The Quarterly Journal of Economics*, 130(4): 1885–1939.
- Asunka, Joseph, Sarah Brierley, Miriam Golden, Eric Kramon, and George Ofosu. 2017. "Electoral Fraud or Violence: The Effect of Observers on Party Manipulation Strategies." *British Journal of Political Science*, 1–23.
- Banerjee, Abhijit, Esther Duflo, Clement Imbert, Santhosh Mathew, and Rohini Pande. 2020. "E-governance, Accountability, and Leakage in Public Programs: Experimental Evidence from a Financial Management Reform in India." American Economic Journal: Applied Economics.
- Banerjee, Abhijit V, Esther Duflo, and Rachel Glennerster. 2008. "Putting a Band-Aid on a Corpse: Incentives for Nurses in the Indian Public Health Care System." Journal of the European Economic Association, 6(2-3): 487–500.
- **Besley, Timothy.** 2006. Principled Agents? The Political Economy of Good Government. Oxford University Press on Demand.
- Besley, Timothy, and Torsten Persson. 2011. *Pillars of Prosperity*. Princeton University Press.
- Bossuroy, Thomas, Clara Delavallade, and Vincent Pons. 2019. "Biometric Tracking, Healthcare Provision, and Data Quality: Experimental Evidence from Tuberculosis Control." *NBER Working Paper*.
- Bunce, Valerie J, and Sharon L Wolchik. 2011. Defeating Authoritarian Leaders in Postcommunist Countries. Cambridge University Press.
- Callen, Michael, and James D Long. 2015. "Institutional Corruption and Election Fraud: Evidence from a Field Experiment in Afghanistan." *American Economic Review*, 105(1): 354–81.
- Calonico, Sebastian, Matias D Cattaneo, and Rocio Titiunik. 2014. "Robust Nonparametric Confidence Intervals for Regression-Discontinuity Designs." *Econometrica*, 82(6): 2295–2326.

- Cantoni, Davide, Yuyu Chen, David Y Yang, Noam Yuchtman, and Y Jane Zhang. 2017. "Curriculum and Ideology." Journal of Political Economy, 125(2): 338– 392.
- Casas, Agustin, Guillermo Diaz, and Andre Trindade. 2017. "Who Monitors the Monitor? Effect of Party Observers on Electoral Outcomes." *Journal of Public Economics*, 145: 136–149.
- Chen, Yuyu, and David Y Yang. 2019. "The Impact of Media Censorship: 1984 or Brave New World?" American Economic Review, 109(6): 2294–2332.
- Cox, Gary W. 2009. "Authoritarian Elections and Leadership Succession, 1975-2004." APSA 2009 Toronto Meeting Paper.
- Cruz, Cesi, Philip Keefer, and Carlos Scartascini. 2021. "Database of Political Institutions 2020: Codebook." Washington, DC: Inter-American Development Bank.
- **Duflo, Esther, Rema Hanna, and Stephen P Ryan.** 2012. "Incentives Work: Getting Teachers to Come to School." *American Economic Review*, 102(4): 1241–78.
- Egorov, Georgy, and Konstantin Sonin. 2020. "The Political Economics of Non-Democracy." National Bureau of Economic Research.
- Egorov, Georgy, and Konstantin Sonin. 2021. "Elections in Non-democracies." *The Economic Journal*, 131(636): 1682–1716.
- Enikolopov, Ruben, Vasily Korovkin, Maria Petrova, Konstantin Sonin, and Alexei Zakharov. 2013. "Field Experiment Estimate of Electoral Fraud in Russian Parliamentary Elections." *Proceedings of the National Academy of Sciences*, 110(2): 448–452.
- Fearon, James D. 2011. "Self-Enforcing Democracy." The Quarterly Journal of Economics, 126(4): 1661–1708.
- Frye, Timothy, Ora John Reuter, and David Szakonyi. 2014. "Political Machines at Work: Voter Mobilization and Eectoral Subversion in the Workplace." *World Politics*, 66(2): 195–228.
- Frye, Timothy, Ora John Reuter, and David Szakonyi. 2019a. "Hitting Them with Carrots: Voter Intimidation and Vote Buying in Russia." British Journal of Political Science, 49(3): 857–881.
- Frye, Timothy, Ora John Reuter, and David Szakonyi. 2019b. "Vote Brokers, Clientelist Appeals, and Voter Turnout: Evidence from Russia and Venezuela." World Politics, 71(4): 710–746.

- Fujiwara, Thomas. 2015. "Voting Technology, Political Responsiveness, and Infant Health: Evidence from Brazil." *Econometrica*, 83(2): 423–464.
- Gandhi, Jennifer, and Ellen Lust-Okar. 2009. "Elections Under Authoritarianism." Annual Review of Political Science, 12: 403–422.
- Gehlbach, Scott, and Alberto Simpser. 2015. "Electoral Manipulation as Bureaucratic Control." American Journal of Political Science, 59(1): 212–224.
- Gehlbach, Scott, Konstantin Sonin, and Milan W Svolik. 2016. "Formal Models of Nondemocratic Politics." Annual Review of Political Science, 19: 565–584.
- Guriev, Sergei M, and Daniel Treisman. 2019a. "Informational Autocrats." Journal of Economic Perspectives.
- Guriev, Sergei M, and Daniel Treisman. 2019b. "A Theory of Informational Autocracy." Journal of Public Economics.
- Herron, Erik S. 2010. "The Effect of Passive Observation Methods on Azerbaijan's 2008 Presidential Election and 2009 Referendum." *Electoral Studies*, 29(3): 417–424.
- **Hyde, Susan D.** 2007. "The Observer Effect in International Politics: Evidence from a Natural Experiment." *World Politics*, 60(1): 37–63.
- Hyde, Susan D, and Nikolay Marinov. 2012. "Which Elections Can Be Lost?" Political analysis, 20(2): 191–210.
- Hyde, Susan D, and Nikolay Marinov. 2014. "Information and Self-Enforcing Democracy: The Role of International Election Observation." *International Organization*, 68(2): 329–359.
- Ichino, Nahomi, and Matthias Schündeln. 2012. "Deterring or Displacing Electoral Irregularities? Spillover Effects of Observers in a Randomized Field Experiment in Ghana." The Journal of Politics, 74(1): 292–307.
- King, Gary, Jennifer Pan, and Margaret E Roberts. 2017. "How the Chinese Government Fabricates Social Media Posts for Strategic Distraction, Not Engaged Argument." American Political Science Review, 111(3): 484–501.
- Klimek, Peter, Yuri Yegorov, Rudolf Hanel, and Stefan Thurner. 2012. "Statistical Detection of Systematic Election Irregularities." Proceedings of the National Academy of Sciences, 109(41): 16469–16473.
- Kobak, Dmitry, Sergey Shpilkin, and Maxim S Pshenichnikov. 2016. "Statistical Fingerprints of Electoral Fraud?" *Significance*, 13(4): 20–23.

- Lacasa, Lucas, and Juan Fernández-Gracia. 2019. "Election Forensics: Quantitative Methods for Electoral Fraud Detection." *Forensic Science International*, 294: 19–22.
- Lewis-Faupel, Sean, Yusuf Neggers, Benjamin A Olken, and Rohini Pande. 2016. "Can Electronic Procurement Improve Infrastructure Provision? Evidence from Public Works in India and Indonesia." *American Economic Journal: Economic Policy*, 8(3): 258–83.
- Little, Andrew T. 2012. "Elections, Fraud, and Election Monitoring in the Shadow of Revolution." *Quarterly Journal of Political Science*, 7(3): 249–283.
- Lührmann, Anna, and Staffan I Lindberg. 2019. "A Third Wave of Autocratization Is Here: What Is New About It?" *Democratization*, 26(7): 1095–1113.
- Malesky, Edmund, and Paul Schuler. 2011. "The Single-Party Dictator's Dilemma: Information in Elections without Opposition." *Legislative Studies Quarterly*, 36(4): 491–530.
- Martinez-Bravo, Monica, Gerard Padró I Miquel, Nancy Qian, and Yang Yao. 2017. "The Rise and Fall of Local Elections in China: Theory and Empirical Evidence on the Autocrat's Trade-off." *National Bureau of Economic Research*.
- McCrary, Justin. 2008. "Manipulation of the Running Variable in the Regression Discontinuity Design: A Density Test." *Journal of Econometrics*, 142(2): 698–714.
- Miller, Michael K. 2015. "Elections, Information, and Policy Responsiveness in Autocratic Regimes." *Comparative Political Studies*, 48(6): 691–727.
- Muralidharan, Karthik, Paul Niehaus, and Sandip Sukhtankar. 2016. "Building State Capacity: Evidence from Biometric Smartcards in India." *American Economic Review*, 106(10): 2895–2929.
- Muralidharan, Karthik, Paul Niehaus, Sandip Sukhtankar, and Jeffrey Weaver. 2021. "Improving Last-Mile Service Delivery using Phone-Based Monitoring." American Economic Journal: Applied Economics.
- Myagkov, Mikhail G, and Peter C Ordeshook. 2008. Russian Elections: An Oxymoron of Democracy. National Council for Eurasian and East European Research Seattle, WA.
- Myagkov, Mikhail, Peter C. Ordeshook, and Dimitri Shakin. 2009. The Forensics of Election Fraud: Russia and Ukraine. Cambridge University Press.
- **Rozenas, Arturas.** 2016. "Office Insecurity and Electoral Manipulation." *The Journal* of *Politics*, 78(1): 232–248.

- Rundlett, Ashlea, and Milan W Svolik. 2016. "Deliver the Vote! Micromotives and Macrobehavior in Electoral Fraud." American Political Science Review, 110(1): 180– 197.
- Simpser, Alberto. 2013. Why Governments and Parties Manipulate Elections: Theory, Practice, and Implications. Cambridge University Press.
- Sjoberg, Fredrik M. 2014. "Autocratic Adaptation: The Strategic Use of Transparency and The Persistence of Election Fraud." *Electoral Studies*, 33: 233–245.
- **Tucker, Joshua A.** 2007. "Enough! Electoral Fraud, Collective Action Problems, and Post-Communist Colored Revolutions." *Perspectives on Politics*, 5(3): 535–551.
- Wintrobe, Ronald. 2000. The Political Economy of Dictatorship. Cambridge University Press.
- World Bank. 2016. "World Development Report 2016: Digital Dividends." World Bank Publications.
- Wright, Joseph. 2009. "How Foreign Aid Can Foster Democratization in Authoritarian Regimes." *American journal of political science*, 53(3): 552–571.

Appendix

A. Background

Figure A.1: Screenshots of the Live-Stream Platform, www.nashvybor2018.ru



Notes: This figure shows screenshots of the live-stream platform, made during the 2018 local elections. The upper left picture captures a place of issuance of ballots to voters, while the upper right picture captures ballot boxes at the same polling station (Khabarovski Krai, #135). The bottom picture illustrates how monitoring was crowdsourced, showing how a webcam feed could be accessed over the internet for any monitored station.

B. Election Records Data



Figure A.2: Map of Studied Regions of Russia

Notes: The blue color highlights the twenty-six studied regions, which provided information on the location of video monitoring on the websites of regional election commissions.

Regions:	All	Studied	Non-Studied	Difference	P-value
—	(1)	(2)	(3)	(4)	(5)
Panel A: 2018 Presidentia	l Election				
Reported Turnout	0.69	0.68	0.69	-0.01	0.49
	(0.09)	[0.02]	[0.01]	[0.02]	
Incumbent's Vote Share	0.77	0.76	0.77	-0.01	0.53
	(0.07)	[0.01]	[0.01]	[0.01]	
# Observations	85	26	59	85	
Panel B: 2018 Presidentia	l Election (E.	xcluding Crime	a and Sevastopol)	
Reported Turnout	0.69	0.68	0.69	-0.01	0.47
	(0.09)	[0.02]	[0.01]	[0.02]	
Incumbent's Vote Share	0.76	0.75	0.77	-0.01	0.36
	(0.06)	[0.01]	[0.01]	[0.01]	
# Observations	83	25	58	83	
Panel C: 2012 Presidentia	l Election				
Reported Turnout	0.67	0.66	0.67	-0.01	0.83
	(0.10)	[0.02]	[0.01]	[0.02]	
Incumbent's Vote Share	0.64	0.62	0.66	-0.04	0.05
	(0.10)	[0.01]	[0.01]	[0.02]	
# Observations	83	25	58	83	

Table A.1: Regional Summary Statistics

Notes: This table reports regional summary stistics for studied and non-studied regions. Robust standard errors are reported in brackets. Standard deviations are reported in parentheses.

C. Threats to Identification: Discontinuities in Polling Station Characteristics



Figure A.3: Electronic Vote-Count Machines

Notes: This figure plots the share of polling stations with electronic vote-counting machines in bins of 100 registered voters. The red vertical line depicts the cutoff of 1,000 registered voters, which defined eligibility for video monitoring.





Notes: This figure plots the average number of polling station employees in bins of 25 registered voters. The sample includes polling stations in the bandwidth of 400 registered voters in two studied regions of Russia. The red vertical line depicts the cutoff of 1,000 registered voters, which defined eligibility for video monitoring.

D. Impact on Election Outcomes: Additional Results

Outcome:	Incumbent's Vote Share			Incun	Incumbent's Vote Margin		
Sample:	Full	Rural	Urban	Full	Rural	Urban	
	(1)	(2)	(3)	(4)	(5)	(6)	
Second Stage (2SLS)	-0.009*	-0.001	-0.009	-0.012	0.003	-0.012	
	[0.006]	[0.009]	[0.008]	[0.009]	[0.015]	[0.013]	
Outcome Mean	0.755	0.773	0.732	0.621	0.648	0.589	
Region FEs	No	No	No	No	No	No	
Commission FEs	Yes	Yes	Yes	Yes	Yes	Yes	
# Observations	5721	3142	2515	5721	3142	2515	

Table A.2: Effects on Incumbent's Vote Share and Margin of Victory

* p < 0.10,** p < 0.05,*** p < 0.01

Notes: This table reports the effects of video monitoring on the incumbent's vote share and margin of victory in the 2018 presidential election. Specifications use data at the polling station level, exclude intermediate data points in the interval of [975, 1000) registered voters, exclude election commission clusters with a single observation, and include separate linear trends on each cutoff side with triangular weights. The bandwidth is 400 registered voters on both sides of the cutoff. Standard errors clustered by election commission are reported in brackets. Standard deviations are reported in parentheses.

Outcome:	Votes Cast for							
	Cand 1	Cand 2	Cand 3	Cand 4	Cand 5	Cand 6	Cand 7	
Second Stage (ToT)	-0.202	-1.967	-0.623	0.085	0.363	-0.180	0.059	
	[0.302]	[3.288]	[1.542]	[0.624]	[0.317]	[0.279]	[0.442]	
Outcome Mean	4.063	85.192	43.339	7.473	4.479	3.860	4.071	
	(3.001)	(45.289)	(20.834)	(7.460)	(4.303)	(4.276)	(5.043)	
Region FEs	No	No	No	No	No	No	No	
Commission FEs	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
# Observations	5721	5721	5721	5721	5721	5721	5721	

Table A.3: Effects on Votes Cast for Other Candidates

Notes: This table reports the effects of video monitoring on the number of votes cast for other candidates in the 2018 presidential election. Specifications use data at the polling station level, exclude intermediate data points in the interval of [975, 1000) registered voters, exclude election commission clusters with a single observation, and include separate linear trends on each cutoff side with triangular weights. The bandwidth is 400 registered voters on both sides of the cutoff. Standard errors clustered by election commission are reported in brackets. Standard deviations are reported in parentheses.

Outcome:	Share of Polling Stations with Abnormally High ($\geq 80\%$)					
	Turnout	Inc. Vote Share	Turnout and Inc.			
			Vote Share			
	(1)	(2)	(3)			
Reduced Form (ITT)	0.008	0.009	0.008			
	[0.018]	[0.016]	[0.013]			
Outcome Mean	0.121	0.112	0.046			
	(0.326)	(0.316)	(0.210)			
Region FEs	No	No	No			
Commission FEs	Yes	Yes	Yes			
# Observations	5364	5364	5364			

Table A.4: Mechanisms: Placebo Effects on Fraud Indicators, 2012 Presidential Election

Notes: This table reports the placebo effects of video monitoring on the share of polling stations with abnormally high values ($\geq 80\%$) of reported turnout and incumbent vote share, fraud indicators in election forensics literature (Klimek et al., 2012), in the 2012 presidential election. Specifications use data at the polling station level, exclude election commission clusters with a single observation, and include separate linear trends on each cutoff side with triangular weights. The bandwidth is 400 registered voters on both sides of the cutoff. Standard errors clustered by election commission are reported in brackets. Standard deviations are reported in parentheses.

Outcome:	Share of Polling Stations with Rounded Values of						
		Turnout		Inc. Vote Share			
Sample:	Full	Rural	Urban	Full	Rural	Urban	
	(1)	(2)	(3)	(4)	(5)	(6)	
Second Stage (ToT)	0.010	0.039	-0.010	-0.035	-0.002	-0.060	
	[0.037]	[0.056]	[0.052]	[0.032]	[0.047]	[0.048]	
Outcome Mean	0.113	0.117	0.109	0.107	0.113	0.101	
	(0.317)	(0.322)	(0.311)	(0.310)	(0.317)	(0.301)	
Region FEs	No	No	No	No	No	No	
Commission FEs	Yes	Yes	Yes	Yes	Yes	Yes	
# Observations	5721	3142	2515	5721	3142	2515	

Table A.5: Mechanisms: Effects on Vote-Count Fraud Indicators

Notes: This table reports the effects of video monitoring on the share of polling stations with rounded values of reported turnout and incumbent vote share, indicators of vote-count fraud in election forensics literature (Klimek et al., 2012), in the 2018 presidential election. Rounded values are decimal numbers within a 0.05 margin of the nearesest integer number. Specifications use data at the polling station level, exclude intermediate data points in the interval of [975, 1000) registered voters, exclude election commission clusters with a single observation, and include separate linear trends on each cutoff side with triangular weights. The bandwidth is 400 registered voters on both sides of the cutoff. Standard errors clustered by election commission are reported in brackets. Standard deviations are reported in parentheses.



Figure A.5: Effects of Video Monitoring on Reported Turnout over Time

Notes: This figure plots reduced-form estimates of the impact of video monitoring on the reported turnout over time. Each point plots means of observations in bins of 25 registered voters. The solid lines plot predicted values and 95% confidence intervals of a linear regression estimated separately on either side of the threshold within the bandwidth of 400 registered voters.

Outcome:	Reported Turnout						
	10 am	$8 \mathrm{pm}$					
-	(1)	(2)	(3)	(4)	(5)		
Second Stage (ToT)	-0.011***	-0.011	-0.018**	-0.027***	-0.035***		
	[0.004]	[0.007]	[0.008]	[0.009]	[0.009]		
Outcome Mean	0.102	0.291	0.496	0.607	0.677		
	(0.053)	(0.090)	(0.104)	(0.114)	(0.124)		
Region FEs	No	No	No	No	No		
Commission FEs	Yes	Yes	Yes	Yes	Yes		
# Observations	5721	5721	5721	5721	5721		

Table A.6: Mechanisms: Effects on Reported Turnout Over Time

Notes: This table reports the effects of video monitoring on the reported turnout over time. Specifications use data at the polling station level, exclude intermediate data points in the interval of [975, 1000) registered voters, exclude election commission clusters with a single observation, and include separate linear trends on each cutoff side with triangular weights. The bandwidth is 400 registered voters on both sides of the cutoff. Standard errors clustered by election commission are reported in brackets. Standard deviations are reported in parentheses.

Sample:	Full	2012 Winner	s's Vote Share	2008 Winner	s's Vote Share			
-		High	Low	High	Low			
	(1)	(2)	(3)	(4)	(5)			
Panel A: Share of Video-Monitored Polling Stations								
First Stage	0.595^{***}	0.579^{***}	0.609^{***}	0.540^{***}	0.669^{***}			
	[0.028]	[0.037]	[0.044]	[0.037]	[0.041]			
Outcome Mean	0.436	0.391	0.492	0.379	0.510			
	(0.496)	(0.488)	(0.500)	(0.485)	(0.500)			
Panel B: Effect on R	eported Turne	out						
Second Stage (ToT)	-0.035***	-0.057***	-0.010	-0.046***	-0.024*			
	[0.009]	[0.012]	[0.012]	[0.013]	[0.012]			
Outcome Mean	0.677	0.705	0.643	0.714	0.630			
	(0.124)	(0.133)	(0.102)	(0.130)	(0.098)			
Panel C: Effect on In	ncumbent's Va	otes						
Second Stage (ToT)	-41.572***	-57.726***	-23.303*	-56.691^{***}	-25.887**			
	[9.357]	[13.213]	[12.910]	[14.323]	[11.826]			
Outcome Mean	502.540	517.484	481.305	532.810	460.845			
	(165.244)	(175.722)	(146.273)	(175.665)	(137.897)			
Panel D: Effect on O	thers' Votes							
Second Stage (ToT)	-2.465	-8.288	3.773	-4.563	-0.354			
	[4.794]	[6.039]	[7.767]	[6.076]	[7.657]			
Outcome Mean	152.477	139.789	169.160	139.531	170.229			
	(66.269)	(66.557)	(61.851)	(66.350)	(61.669)			
Region FEs	No	No	No	No	No			
Commission FEs	Yes	Yes	Yes	Yes	Yes			
# Observations	5721	3123	2561	3186	2498			

Table A.7: Mechanisms: Heterogeneous Effects by Winner's Vote Share in the 2012 and 2008 Presidential Elections

Notes: This table reports the heterogeneous effects of video monitoring on voting by the winner's vote share in the 2012 and 2008 presidential elections. The high (low) winner's vote share is defined by the above (below) median value at the election commission level. Specifications use data at the polling station level, exclude intermediate data points in the interval of [975, 1000) registered voters, exclude election commission clusters with a single observation, and include separate linear trends on each cutoff side with triangular weights. The bandwidth is 400 registered voters on both sides of the cutoff. Standard errors clustered by election commission are reported in brackets. Standard deviations are reported in parentheses.

E. Impact on Voting: Placebo Effects



Figure A.6: Placebo Effects on Voting, 2012 Presidential Election

Notes: This figure plots reduced-form estimates of the impact of video monitoring on voting in the 2012 presidential election. Each point plots means of observations in bins of 25 registered voters. The solid lines plot predicted values and 95% confidence intervals of a linear regression estimated separately on either side of the threshold within the bandwidth of 400 registered voters. For presentation purposes, detrended graphs difference out trends associated with a positive relationship between the number of votes and registered voters.

	(1)	(2)	(3)
Panel A: Effect on Rep	orted Turnout		
Reduced Form (ITT)	0.010	0.006	0.008
	[0.007]	[0.006]	[0.005]
Outcome Mean	0.658	0.658	0.658
	(0.115)	(0.115)	(0.115)
Panel B: Effect on Incu	umbent's Votes		
Reduced Form (ITT)	7.762	5.260	7.470
	[7.659]	[6.274]	[5.080]
Outcome Mean	412.566	412.566	412.566
	(141.323)	(141.323)	(141.323)
Panel C: Effect on Othe	ers' Votes		
Reduced Form (ITT)	0.581	-0.311	-0.793
	[3.759]	[3.158]	[2.796]
Outcome Mean	223.596	223.596	223.596
	(87.819)	(87.819)	(87.819)
Region FEs	No	Yes	No
Commission FEs	No	No	Yes
# Observations	5364	5364	5364

Table A.8: Placebo Effects on Voting, 2012 Presidential Election

Notes: This table reports the placebo effects of video monitoring on voting in the 2012 presidential election. Specifications use data at the polling station level, exclude election commission clusters with a single observation, and include separate linear trends on each cutoff side with triangular weights. The bandwidth is 400 registered voters on both sides of the cutoff. Standard errors clustered by election commission are reported in brackets. Standard deviations are reported in parentheses.

Cutoff:	1,000	850	$1,\!150$
-	(1)	(2)	(3)
Panel A: Share of Vide	eo-Monitored Pa	olling Stations	
First Stage	0.595^{***}	-0.074***	-0.136***
	[0.028]	[0.017]	[0.022]
Outcome Mean	0.436	0.256	0.616
	(0.496)	(0.436)	(0.486)
Panel B: Effect on Tur	rnout		
Reduced Form (ITT)	-0.021***	-0.001	0.002
	[0.005]	[0.005]	[0.005]
Second Stage (ToT)	-0.035***	0.013	-0.018
	[0.009]	[0.070]	[0.037]
Outcome Mean	0.677	0.693	0.662
	(0.124)	(0.130)	(0.117)
Panel C: Effect on Inc	umbent's Votes		
Reduced Form (ITT)	-24.717***	1.025	-1.361
	[5.666]	[4.727]	[6.249]
Second Stage (ToT)	-41.572***	-13.935	9.976
	[9.357]	[63.915]	[46.082]
Outcome Mean	502.540	424.082	571.070
	(165.244)	(155.865)	(172.590)
Panel D: Effect on Oth	ners' Votes		
Reduced Form (ITT)	-1.465	1.754	-4.608
	[2.847]	[2.348]	[2.993]
Second Stage (ToT)	-2.465	-23.859	33.779
	[4.794]	[31.945]	[22.483]
Outcome Mean	152.477	123.131	179.947
	(66.269)	(61.029)	(68.706)
Region FEs	No	No	No
Commission FEs	Yes	Yes	Yes
# Observations	5721	6515	5764

Table A.9: Placebo Effects on Voting, Alternative Cutoffs

Notes: This table reports the placebo effects of video monitoring on voting using alternative cutoffs of 850 and 1,150 registered voters. Specifications use data at the polling station level, exclude intermediate data points in the interval of [975, 1000) registered voters for the cutoff of 1,000 registered voters, exclude election commission clusters with a single observation, and include separate linear trends on each cutoff side with triangular weights. The bandwidth is 400 registered voters on both sides of the cutoff. Standard errors clustered by election commission are reported in brackets. Standard deviations are reported in parentheses.

F. Impact on Election Outcomes: Robustness of Results

Sample:	Full	Full	Full	Rural	Urban				
-	(1)	(2)	(3)	(4)	(5)				
Panel A: Share of Video-Monitored Polling Stations									
First Stage	0.488^{***}	0.489^{***}	0.484^{***}	0.460^{***}	0.481^{***}				
	[0.027]	[0.027]	[0.027]	[0.036]	[0.045]				
Outcome Mean	0.437	0.437	0.437	0.256	0.663				
	(0.496)	(0.496)	(0.496)	(0.436)	(0.473)				
Panel B: Effect on Rep	ported Turnov	ut							
Reduced Form (ITT)	-0.010	-0.012**	-0.015***	-0.021***	-0.004				
	[0.008]	[0.006]	[0.005]	[0.006]	[0.006]				
Second Stage (2SLS)	-0.021	-0.024**	-0.030***	-0.045***	-0.009				
	[0.016]	[0.012]	[0.009]	[0.014]	[0.013]				
Outcome Mean	0.677	0.677	0.677	0.704	0.644				
	(0.124)	(0.124)	(0.124)	(0.127)	(0.110)				
Panel C: Effect on Inc	umbent's Vot	tes							
Reduced Form (ITT)	-12.344	-13.318**	-17.111***	-17.355**	-10.432				
	[8.726]	[6.183]	[4.887]	[6.935]	[7.069]				
Second Stage (2SLS)	-25.279	-27.247**	-35.362***	-37.702**	-21.686				
	[17.748]	[12.407]	[9.905]	[14.838]	[14.497]				
Outcome Mean	502.010	502.010	502.010	489.247	518.772				
	(164.383)	(164.383)	(164.383)	(166.148)	(160.609)				
Panel D: Effect on Oth	ners' Votes								
Reduced Form (ITT)	-1.046	-1.285	-1.041	-4.525	0.560				
	[3.144]	[2.666]	[2.400]	[3.093]	[4.273]				
Second Stage (2SLS)	-2.142	-2.630	-2.152	-9.830	1.164				
	[6.424]	[5.458]	[4.958]	[6.761]	[8.875]				
Outcome Mean	152.364	152.364	152.364	132.122	177.354				
	(65.669)	(65.669)	(65.669)	(58.838)	(65.260)				
Region FEs	No	Yes	No	No	No				
Commission FEs	No	No	Yes	Yes	Yes				
# Observations	5914	5914	5914	3244	2609				

Table A.10: Robustness of Effects on Voting to Inclusion of Intermediate Data Points (975–1,000 Registered Voters)

* p < 0.10, ** p < 0.05, *** p < 0.01

Notes: This table reports the robustness of effects on voting to the inclusion of intermediate data points with 975–1,000 registered voters. Specifications use data at the polling station level, exclude election commission clusters with a single observation, and include separate linear trends on each cutoff side with triangular weights. The bandwidth is 400 registered voters on both sides of the cutoff. Standard errors clustered by election commission are reported in brackets. Standard deviations are reported in parentheses.

Functional Form:	Linear	Quadratic	Qubic	Local Linear					
-	(1)	(2)	(3)	(4)					
Panel A: Share of Video-Monitored Polling Stations									
First Stage	0.595^{***}	0.481^{***}	0.399^{***}	0.598^{***}					
	[0.028]	[0.045]	[0.069]	[0.030]					
Outcome Mean	0.436	0.436	0.436	0.436					
	(0.496)	(0.496)	(0.496)	(0.496)					
Panel B: Effect on Rep	orted Turnout								
Reduced Form (ITT)	-0.021***	-0.026***	-0.038***	-0.021**					
	[0.005]	[0.009]	[0.014]	[0.009]					
Second Stage (ToT)	-0.035***	-0.054***	-0.097***	-0.035**					
	[0.009]	[0.018]	[0.037]	[0.014]					
Outcome Mean	0.677	0.677	0.677	0.677					
	(0.124)	(0.124)	(0.124)	(0.124)					
Panel C: Effect on Incu	umbent's Votes								
Reduced Form (ITT)	-24.717***	-22.947**	-27.601*	-23.611***					
	[5.666]	[9.399]	[14.590]	[8.915]					
Second Stage (ToT)	-41.572***	-47.657**	-69.258*	-39.477***					
	[9.357]	[19.318]	[37.488]	[14.620]					
Outcome Mean	502.540	502.540	502.540	502.540					
	(165.244)	(165.244)	(165.244)	(165.244)					
Panel D: Effect on Oth	ers' Votes								
Reduced Form (ITT)	-1.465	-8.669*	-10.038	-2.352					
	[2.847]	[4.755]	[7.331]	[3.390]					
Second Stage (ToT)	-2.465	-18.005*	-25.188	-3.933					
	[4.794]	[10.074]	[19.109]	[5.681]					
Outcome Mean	152.477	152.477	152.477	152.477					
	(66.269)	(66.269)	(66.269)	(66.269)					
Region FEs	No	No	No	Yes					
Commission FEs	Yes	Yes	Yes	No					
# Observations	5721	5721	5721	5721					

Table A.11: Robustness of Effects on Voting to Alternative Functional Forms

Notes: This table reports the robustness of effects on voting to alternative functional forms. Specifications use data at the polling station level, exclude intermediate data points in the interval of [975, 1000) registered voters, exclude election commission clusters with a single observation, and include separate functions on each cutoff side with triangular weights. The bandwidth is 400 registered voters on both sides of the cutoff. Standard errors clustered by election commission are reported in brackets. Standard deviations are reported in parentheses.

Bandwidth:	400	300	500	CCT	CCT	CCT			
	(1)	(2)	(3)	(4)	(5)	(6)			
Panel A: Share of Vid	Panel A: Share of Video-Monitored Polling Stations								
First Stage	0.595^{***}	0.550^{***}	0.622^{***}	0.498^{***}	0.565^{***}	0.475^{***}			
	[0.028]	[0.034]	[0.024]	[0.043]	[0.032]	[0.049]			
Outcome Mean	0.436	0.453	0.419	0.464	0.446	0.471			
	(0.496)	(0.498)	(0.493)	(0.499)	(0.497)	(0.499)			
Panel B: Effect on Re	ported Turnov	ıt							
Reduced Form (ITT)	-0.021***	-0.023***	-0.021***	-0.025***	-0.022***	-0.027***			
· · · · · · · · · · · · · · · · · · ·	[0.005]	[0.007]	[0.005]	[0.008]	[0.006]	[0.009]			
Second Stage (ToT)	-0.035***	-0.042***	-0.034***	-0.050***					
<u> </u>	[0.009]	[0.012]	[0.007]	[0.017]					
Outcome Mean	0.677	0.675	0.679	0.674					
	(0.124)	(0.125)	(0.125)	(0.124)					
Panel C: Effect on Inc	cumbent's Vot	es							
Reduced Form (ITT)	-24.717***	-24.161***	-25.820***	-23.555**	-23.839***	-24.439**			
· · ·	[5.666]	[7.036]	[4.947]	[9.190]	[6.535]	[10.597]			
Second Stage (ToT)	-41.572***	-43.936***	-41.513***		-42.157***				
	[9.357]	[12.540]	[7.850]		[11.331]				
Outcome Mean	502.540	506.617	492.012		504.551				
	(165.244)	(152.813)	(181.102)		(156.120)				
Panel D: Effect on Ot	hers' Votes								
Reduced Form (ITT)	-1.465	-4.234	-0.195	-7.029	-3.302	-7.393			
· · ·	[2.847]	[3.578]	[2.477]	[4.751]	[3.310]	[5.622]			
Second Stage (ToT)	-2.465	-7.699	-0.313			-15.567			
- 、 ,	[4.794]	[6.538]	[3.983]			[11.975]			
Outcome Mean	152.477	153.747	149.535			154.815			
	(66.269)	(63.114)	(71.697)			(59.095)			
Region FEs	No	No	No	No	No	No			
Commission FEs	Yes	Yes	Yes	Yes	Yes	Yes			
# Observations	5721	4040	7599	2812	4522	2352			

Table A.12: Robustness of Effects on Voting to Alternative Bandwidths

Notes: This table reports the robustness of effects on voting to alternative bandwidths. Specifications use data at the polling station level, exclude intermediate data points in the interval of [975, 1000) registered voters, exclude election commission clusters with a single observation, and include separate linear trends on each cutoff side with triangular weights. The bandwidth is specified in each column; CCT corresponds to the optimal bandwidth, computed separately for each outcome using the procedures by Calonico, Cattaneo and Titiunik (2014). Standard errors clustered by election commission are reported in brackets. Standard deviations are reported in parentheses.

G. Displacement Effects



Figure A.7: Estimating Vote Displacement, Selection of Polling Stations





Notes: This figure shows examples of polling stations included and excluded from the analysis of displacement effects in the radius of 5 kilometers. Panel A shows an example of the included unmonitored polling station, which has one neighbor with 1,000–1,400 voters. Panel B shows an example of the included unmonitored polling station, which has two neighbors with 600-1,000 registered voters. Panel C shows an example of excluded unmonitored polling station because it has a neighbor with more than 1,400 voters.

Radius:	3]	km	5	km	7]	km	
-	(1)	(2)	(3)	(4)	(5)	(6)	
Panel A: Share of Video-Monitored Neighboring Polling Stations							
First Stage	0.586^{***}	0.594^{***}	0.646^{***}	0.650^{***}	0.636^{***}	0.628^{***}	
	[0.077]	[0.064]	[0.072]	[0.058]	[0.064]	[0.058]	
Outcome Mean	0.192	0.192	0.187	0.187	0.164	0.164	
	(0.394)	(0.394)	(0.390)	(0.390)	(0.371)	(0.371)	
Panel B: Effect on Rep	orted Turnov	ut					
Reduced Form (ITT)	0.042	0.013	0.018	0.011	0.025	0.022	
	[0.028]	[0.016]	[0.026]	[0.014]	[0.023]	[0.015]	
Second Stage (2SLS)	0.072	0.022	0.028	0.017	0.039	0.036	
	[0.050]	[0.027]	[0.041]	[0.022]	[0.037]	[0.024]	
Outcome Mean	0.687	0.687	0.698	0.698	0.712	0.712	
	(0.126)	(0.126)	(0.126)	(0.126)	(0.130)	(0.130)	
Panel C: Effect on Inc	umbent's Vot	e Share					
Reduced Form (ITT)	0.036	0.024^{**}	0.027	0.030***	0.036^{**}	0.035^{***}	
	[0.022]	[0.011]	[0.021]	[0.011]	[0.018]	[0.010]	
Second Stage (2SLS)	0.061	0.040**	0.042	0.047^{***}	0.057^{**}	0.056^{***}	
	[0.038]	[0.018]	[0.033]	[0.018]	[0.028]	[0.016]	
Outcome Mean	0.746	0.746	0.755	0.755	0.766	0.766	
	(0.091)	(0.091)	(0.095)	(0.095)	(0.097)	(0.097)	
Panel D: Effect on Oth	ers' Vote Sh	are					
Reduced Form (ITT)	-0.035*	-0.023**	-0.026	-0.029***	-0.035**	-0.034***	
	[0.021]	[0.011]	[0.021]	[0.011]	[0.017]	[0.010]	
Second Stage (2SLS)	-0.060*	-0.039**	-0.040	-0.045**	-0.055**	-0.054***	
	[0.036]	[0.018]	[0.032]	[0.017]	[0.027]	[0.016]	
Outcome Mean	0.242	0.242	0.233	0.233	0.224	0.224	
	(0.088)	(0.088)	(0.092)	(0.092)	(0.094)	(0.094)	
Region FEs	No	Yes	No	Yes	No	Yes	
Commission FEs	No	No	No	No	No	No	
# Observations	2667	2667	3414	3414	4146	4146	

 Table A.13: Effects on Vote Displacement, Robustness to Pooling Neighbors

Notes: This table reports the robustness of vote displacement effects to pooling neighboring polling stations instead of taking means. This estimation (i) excludes unmonitored stations, which have neighbors with more than 1,400 voters (upper bandwidth boundary); (ii) excludes neighbors with fewer than 600 registered voters (lower bandwidth boundary); (iii) excludes neighbors outside of the specified radius from unmonitored polling stations; (iv) restricts the sample to unmonitored polling places, which have neighbors on one side of the cutoff; (v) pools all neighbors. Specifications use data at the polling station level, exclude intermediate data points in the interval of [975, 1000) registered voters, and include separate linear trends on each cutoff side with triangular weights. The bandwidth is 400 registered voters on both sides of the cutoff. Standard errors clustered by election commission are reported in brackets. Standard deviations are reported in parentheses.

Radius:	3 km		5 km		$7 \mathrm{~km}$	
-	(1)	(2)	(3)	(4)	(5)	(6)
Panel A: Share of Video-Monitored Neighboring Polling Stations						
First Stage	0.499***	0.518^{***}	0.561^{***}	0.575^{***}	0.603^{***}	0.609^{***}
	[0.085]	[0.086]	[0.078]	[0.076]	[0.066]	[0.064]
Outcome Mean	0.194	0.194	0.173	0.173	0.167	0.167
	(0.396)	(0.396)	(0.379)	(0.379)	(0.373)	(0.373)
Panel B: Effect on Reported Turnout						
Reduced Form (ITT)	-0.001	0.005	-0.001	0.011	0.004	0.000
	[0.026]	[0.020]	[0.023]	[0.017]	[0.021]	[0.016]
Second Stage (2SLS)	-0.002	0.009	-0.002	0.019	0.006	0.000
- 、 ,	[0.053]	[0.038]	[0.041]	[0.030]	[0.035]	[0.026]
Outcome Mean	0.731	0.731	0.743	0.743	0.752	0.752
	(0.133)	(0.133)	(0.130)	(0.130)	(0.130)	(0.130)
Panel C: Effect on Incumbent's Votes Share						
Reduced Form (ITT)	0.022	0.020	0.019	0.027**	0.020	0.020**
	[0.020]	[0.013]	[0.015]	[0.012]	[0.013]	[0.010]
Second Stage (2SLS)	0.044	0.039	0.034	0.047**	0.034	0.033^{*}
	[0.040]	[0.025]	[0.028]	[0.022]	[0.022]	[0.017]
Outcome Mean	0.782	0.782	0.796	0.796	0.800	0.800
	(0.091)	(0.091)	(0.088)	(0.088)	(0.086)	(0.086)
Panel D: Effect on Others' Vote Share						
Reduced Form (ITT)	-0.022	-0.020	-0.019	-0.026**	-0.021	-0.020**
	[0.019]	[0.013]	[0.015]	[0.012]	[0.013]	[0.010]
Second Stage (2SLS)	-0.044	-0.039	-0.034	-0.046**	-0.035	-0.034**
	[0.038]	[0.026]	[0.027]	[0.022]	[0.022]	[0.017]
Outcome Mean	0.206	0.206	0.194	0.194	0.191	0.191
	(0.087)	(0.087)	(0.085)	(0.085)	(0.084)	(0.084)
Region FEs	No	Yes	No	Yes	No	Yes
Commission FEs	No	No	No	No	No	No
# Observations	727	727	1063	1063	1433	1433

Table A.14: Effects on Vote Displacement, Robustness to a Single Neighbor

Notes: This table reports the robustness of vote displacement effects to a single neighbor restriction. This estimation (i) excludes unmonitored stations, which have neighbors with more than 1,400 voters (upper bandwidth boundary); (ii) excludes neighbors with fewer than 600 registered voters (lower bandwidth boundary); (iii) excludes neighbors outside of the specified radius from unmonitored polling stations; (iv) restricts the sample to unmonitored polling places, which have a single neighbor on either side of the cutoff. Specifications use data at the polling station level, exclude intermediate data points in the interval of [975, 1000) registered voters, and include separate linear trends on each cutoff side with triangular weights. The bandwidth is 400 registered voters on both sides of the cutoff. Standard errors clustered by election commission are reported in brackets. Standard deviations are reported in parentheses.
H. Accountability Effects: Additional Results



Figure A.8: Intensity of Video Monitoring across Districts

Notes: This figure plots the histogram of the share of video-monitored polling stations in bins of 5 percentage points. The unit of observation is a district (rayon). The district is urban if more than half of its population lives in urban areas.

Sector:	All	Administrative	Infrastructure	Education	Social Services	Other
	(1)	(2)	(3)	(4)	(5)	(6)
Panel A: Change in th	he Number o	f Projects				
Second Stage (ToT)	-0.027	-0.006	0.020	-0.031	-0.025**	0.005^{*}
	[0.034]	[0.017]	[0.012]	[0.019]	[0.011]	[0.003]
Outcome Mean	-0.992	0.237	0.180	-0.884	-0.493	0.087
	(10.860)	(5.582)	(3.826)	(6.198)	(3.530)	(0.803)
Panel B: Change in th	he Total Vali	ue of Projects per	Capita			
Second Stage (ToT)	-1.814	-1.284	-0.019	-0.048	-0.098	-0.002
	[1.295]	[0.936]	[0.403]	[0.292]	[0.092]	[0.018]
Outcome Mean	0.756	-8.025	8.176	1.262	-2.478	0.828
	(406.898)	(295.031)	(127.530)	(93.338)	(29.013)	(5.755)
Region FEs	Yes	Yes	Yes	Yes	Yes	Yes
# Observations	596	596	596	596	596	596

Table A.15: Effects on Changes in Public Goods Spending by Economic Sector

* p < 0.10, ** p < 0.05, *** p < 0.01

Notes: This table reports the heterogeneous effects of video monitoring on changes in district public goods spending before the election by the economic sector. Changes in the number of projects and their total value are defined as the differences between December 2017 – March 2018 (four months before the election) and December 2016 – March 2017 (a corresponding period a year earlier), with winsorized values in the bottom- and top-2%. Specifications use data at the district level, instrument the share of video-monitored polling stations with the percentage of polling places above the cutoff in the bandwidth of 400 registered voters, and control for the share of polling stations in the bandwidth. Standard errors clustered by region are reported in brackets. Standard deviations are reported in parentheses.

I. Survey Experiment: Additional Results

Statistic:	Mean Full	Mean C	Control	Mean Priming	Treatment	P-value	# Obs.
		No Sens.	Sens.	No Sens.	Sens.		
		Item	Item	Item	Item		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Geographical Area							
Urban $(=1)$	0.78	0.79	0.77	0.76	0.78	0.86	1097
	(0.42)	[0.05]	[0.05]	[0.05]	[0.05]		
Moscow (=1)	0.13	0.13	0.14	0.10	0.15	0.66	1097
	(0.33)	[0.11]	[0.12]	[0.09]	[0.13]		
St. Petersburg $(=1)$	0.05	0.04	0.05	0.07	0.05	0.75	1097
	(0.22)	[0.04]	[0.04]	[0.07]	[0.05]		
Rural $(=1)$	0.22	0.21	0.23	0.24	0.22	0.86	1097
	(0.42)	[0.05]	[0.05]	[0.05]	[0.05]		
Individual Characteristics							
Female $(=1)$	0.56	0.56	0.57	0.58	0.52	0.48	1097
	(0.50)	[0.03]	[0.02]	[0.02]	[0.03]		
Age	46.21	44.65	46.41	47.94	45.79	0.16	1097
	(16.50)	[1.07]	[0.99]	[1.15]	[0.93]		
Incomplete Secondary School $(=1)$	0.05	0.06	0.03	0.04	0.05	0.46	1097
	(0.21)	[0.02]	[0.01]	[0.01]	[0.01]		
Secondary or Vocational Education $(=1)$	0.64	0.63	0.66	0.62	0.66	0.67	1097
	(0.48)	[0.04]	[0.04]	[0.03]	[0.03]		
Higher Education $(=1)$	0.31	0.31	0.31	0.34	0.29	0.66	1097
3	(0.46)	[0.04]	[0.04]	[0.04]	[0.03]		
Employed $(=1)$	0.60	0.59	0.58	0.55	0.66	0.07	1097
1 5 ()	(0.49)	[0.04]	[0.03]	[0.04]	[0.03]		
Retired $(=1)$	0.26	0.24	0.26	0.32	0.20	0.01	1097
	(0.44)	[0.03]	[0.03]	[0.03]	[0.02]		
Income Level	3.18	3.19	3.15	3.19	3.21	0.92	1096
	(1.02)	[0.08]	[0 09]	[0.07]	[0.09]	0.0-	
Media Consumption	(===)	[0.00]	[0100]	[0.0.1]	[0.00]		
Daily Internet User $(=1)$	0.60	0.64	0.61	0.55	0.62	0.23	1097
Daily morner over (-1)	(0.49)	[0.04]	[0.03]	[0.03]	[0.05]	0.20	1001
Internet Non-User (=1)	0.22	0.19	0.24	0.23	0.20	0.48	1097
internet from ober (-1)	(0.41)	[0.03]	[0.03]	[0.03]	[0.04]	0.10	1001
Daily TV User $(=1)$	0.67	0.67	0.65	0.67	0.70	0.80	1097
Dury IV Ober (-I)	(0.47)	[0.03]	[0.04]	[0.03]	[0.03]	0.00	1001
TV Non User (-1)	0.08	0.08	0.09	0.00	0.07	0.88	1007
	(0.28)	[0.02]	[0.02]	[0.02]	[0.01]	0.00	1007
Government Approval	(0.20)	[0.02]	[0.02]	[0.02]	[0.01]		
Approves President's Work (-1)	0.66	0.66	0.62	0.68	0.71	0.11	1080
Approves i resident s work (-1)	(0.47)	[0.03]	[0.04]	[0.03]	[0.03]	0.11	1000
Approves State Duma's Work (-1)	0.38	0.38	0.36	[0.05]	[0.05]	0.45	1066
Approves State Dunia's work (-1)	(0.38)	[0.04]	[0.03]	[0.04]	[0.03]	0.45	1000
Approves Covernment's Work (-1)	0.45	0.46	0.44	0.45	0.46	0.04	1074
Approves Government's work (-1)	(0.50)	[0.04]	[0.02]	[0.04]	[0.02]	0.94	1074
Approved Appr (-1)	(0.50)	[0.04]	0.67	[0.04]	[0.03]	0.28	1004
Approves Any (=1)	(0.71)	0.74	10.07	0.09	0.75	0.28	1094
The management of Teacher also	(0.40)	[0.04]	[0.05]	[0.05]	[0.05]		
Among of Transport Dellet Derry (1)	0 56	0 56	0.57	0 50	0.60	0.25	1007
Aware of framsparent Ballot Boxes $(=1)$	0.50	06.0	0.07	0.02	00.0	0.35	1097
Among of Video Moritoria - (1)	(0.50)	[0.03]	[0.04]	[0.03]	[0.03]		500
Aware of video Monitoring $(=1)$	0.60	-	-	86.0	0.62	-	538
	(0.49)			[0.03]	[0.04]		500
will Observe Election Online $(=1)$	0.15	-	-	0.17	0.14	-	523
	(0.36)			[0.03]	[0.02]		

Table A.16: Survey Experiment: Summary Statistics and Full Balance Test

Notes: This table reports summary statistics and balance test of random assignment of 1,097 respondents to priming and list experiment treatments in the survey experiment. Column (1) reports the mean of each variable, with standard deviations in parentheses, for the full sample. Columns (2)-(5) report the mean of each variable, with standard deviations. Column (6) reports the p-value of an F-test of joint equality of means between four experimental groups.



Figure A.9: Survey Experiment: Effects of Priming, Excluding Moscow

Notes: This figure plots the effects of the priming treatment, which provides information about video monitoring. The sample consists of 958 respondents who resided outside Moscow and answered that there would be elections in their locality. Each subfigure plots the effects for the full sample (All), the subsample of respondents who were not aware of another transparency tool, transparent ballot boxes (Unaware), and the subsample of respondents who were aware of transparent ballot boxes (Aware). Brackets denote 90% confidence intervals using standard errors clustered by region.

Moscow	
Excluding	
nd Attitudes.	
Voting al	
Priming on	C
: Effects of	
Experiment	
7: Survev	
Table A.1	

Outcome.			ΠV					IInouro	0				A more A		
Outcourse.	Control	Treatment	D turbur	$\Gamma(ahow)_{\alpha}$	MILT	Control	Tweetmant	D moluo	Lebow's	MILT	Control	Tweetmant	D violue	Dichowle	MUT
	Mean	Effect	antea- t	P-value	Q-value	Mean	Effect	anna - 1	P-value	Q-value	Mean	Effect	antex- T	P-value	Q-value
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)	(13)	(14)	(15)
Voting/Supporting United Russia															
Will Vote (=1)	0.61	0.05	0.17	0.15	.78	0.50	0.13^{***}	0.01	0.01	.03	0.68	-0.02	0.69	0.71	1
	(0.49)	[0.03]				(0.50)	[0.05]				(0.47)	[0.04]			
Will Vote for United Russia $(=1)$	0.39	0.01	0.79	0.77	.78	0.27	0.09	0.15	0.12	.18	0.46	-0.03	0.64	0.61	1
	(0.49)	[0.04]				(0.45)	[0.02]				(0.50)	[0.06]			
% Others Voting for United Russia	48.77	1.71	0.29	0.33	.78	45.38	1.79	0.43	0.50	ci.	51.05	1.77	0.41	0.45	1
	(24.47)	[1.61]				(22.13)	[2.24]				(25.72)	[2.13]			
Trust in Election															
Trust in Election Results $(=1)$	0.29	0.06^{**}	0.03	0.05	.07	0.22	0.08^{**}	0.04	0.06	.09	0.34	0.04	0.20	0.30	.68
	(0.45)	[0.03]				(0.42)	[0.04]				(0.47)	[0.03]			
Election Will Lead to Improvements $(=1)$	0.38	-0.02	0.43	0.48	.28	0.33	-0.02	0.61	0.66	.44	0.42	-0.02	0.53	0.58	.68
	(0.49)	[0.03]				(0.47)	[0.04]				(0.49)	[0.04]			
Democratic Values															
Russia is a Democracy $(=1)$	0.83	-0.05*	0.07	0.07	.27	0.82	-0.06	0.15	0.14	.37	0.84	-0.04	0.25	0.29	1
	(0.38)	[0.03]				(0.38)	[0.04]				(0.37)	[0.03]			
# Democratic Values Agreed	2.27	-0.03	0.77	0.79	1	2.36	-0.20	0.18	0.19	.37	2.20	0.10	0.35	0.42	1
	(1.55)	[0.09]				(1.67)	[0.15]				(1.46)	[0.11]			
Tax Compliance is Important $(=1)$	0.66	0.01	0.80	0.85	1	0.64	0.03	0.46	0.53	.37	0.67	-0.01	0.75	0.77	1
	(0.48)	[0.03]				(0.48)	[0.04]				(0.47)	[0.04]			
Indices															
Voting/Supporting United Russia Index	0.00	0.06^{*}	0.10	0.10	.25	0.00	0.18^{***}	0.00	0.00	0	0.00	-0.01	0.91	0.90	76.
	(0.58)	[0.04]				(0.55)	[0.05]				(0.59)	[0.05]			
Trust in Election Index	0.00	0.04	0.34	0.39	.34	0.00	0.08	0.23	0.30	¢.	0.00	0.02	0.71	0.75	76.
	(0.80)	[0.05]				(0.76)	[0.02]				(0.81)	[0.06]			
Democracy Index	0.00	-0.04	0.19	0.23	.34	0.00	-0.07	0.17	0.18		0.00	-0.02	0.69	0.73	26.
	(0.58)	[0.03]				(0.56)	[0.05]				(0.59)	[0.04]			
* $p < 0.10$, ** $p < 0.05$, *** $p < 0.01$ Notes: This table reports the results of the survey i	experiment	based on the r	ore-analysis	plan. Colum	ns (1)–(5) repor	t results for tl	e subsample e	of 958 resno	ndents who	reside outside of	Moscow and a	answered that	there were	zoing to be e	lections in

their locative columns (6)–(10) and (11)–(15) report results separately for subsamples split by averances of transparent balob boxes, which serves as a proxy for averances of transparent balob boxes. We are subsamples split by averances of transparent balob boxes, which serves as a proxy for averances of transparent balob boxes. We are subsamples split by averances of transparent balob boxes, which serves as a proxy for averances of transparent balob boxes. We are subsamples split by averances of transparent balob boxes, which serves as a proxy for averances of transparent balob boxes. We are subsamples split by averances of transparent balob boxes, which serves as a proxy for averances of transparent balob boxes. Columns (1), (6), and (11) report the priming treatment effects of a reminder about video monitoring, with standard errors clustered at the regional level in the treatment effects. Columns (3), (8), and (13) report summer (4), (9), and (14) report transformulation tests of the sharp null hypothesis of no effect from permutation tests with 10,000 repetitions. Columns (3), (10), and (15) report tre-specified q-values which adjust the sampling-based Paules for the false discovery rate (FDR) within each hypothesis group following Anderson (2008) and Yestuleii (2006) and the family-wise error rate (FWER) between indices following Anderson (2008) and Westfall and Young (1993).



Notes: This figure plots the heterogeneous effects of the priming treatment, which provides information about video monitoring. Brackets denote 90% confidence intervals using standard errors clustered by region.



Figure A.11: Survey Experiment: Heterogeneous Effects of Priming (Part 2)

Notes: This figure plots the heterogeneous effects of the priming treatment, which provides information about video monitoring. Brackets denote 90% confidence intervals using standard errors clustered by region.